

SECTION OFFICERS

CHAIR D'Lesli M. Davis Fulbright & Jaworski, L.L.P 2200 Ross Avenue, Suite 2800 2200 H055 Avenue, St Dallas, Texas 75201 Phone (214) 855-8221 Fax (214) 855-8220 ddavie 6 d ddavis@fulbright.com

CHAIR-ELECT Don Valdez Don Valdez Decker, Jones et al. 801 Cherry St, Ste 2000, Unit 46 Fort Worth, Texas 76102-6836 Phone (817) 429-5260 Fax (817) 336-2181 dvaldez@deckerjones.com

SECRETARY Brian Michael Cooper Haynes & Boone LLP 1221 McKinney Street, Suite 2100 Houston, Texas 77010 Phone (713) 547-2060 Fax (713) 236-5444 cooperb@haynesboone.com

TREASURER

Shannon Jamison 3100 Main Street, Ste. 132 Dallas, Texas 75226 Phone (972) 772-9252 Fax (972) 772-3464 jamisonlaw@cebridge.net

IMMEDIATE PAST CHAIR Alan W. Tompkins Unity Hunt, Inc. 1601 Elm Street, Suite 4000 Dallas, Texas 75201 Phone (214) 720-1623 Fax (214) 442-5553 atompkins@unityhunt.com

COUNCIL

Terms Expiring 2010 Tamera P. Bennett Dale A. Head **Rafael Stone**

Terms Expiring 2011 Sally Stacy Craig Crafton Jamey Newberg

Terms Expiring 2012 Will Colgin Catherine Hough Amy E. Mitchell

Ent. Law Institute Director Mike Tolleson

2106 East MLK Blvd. Austin, TX 78702 (512) 480-8822 (512) 479-6212 Fax mike@miketolleson.com

Journal Editor

Sylvester R. Jaime 1011 Highway 6 South, Ste. 216 Houston, TX 77077 (281) 597-9495 (281) 597-9621 Fax srjaimelaw@comcast.net

Webmaster Kenneth W. Pajak 1114 Lost Creek Blvd., Suite 420 Austin, TX 78746 (512) 327-8930 (512) 327-8904 Fax ken@bannerot.com

Texas Entertainment and Sports Law Journal

State Bar of Texas

Entertainment & Sports Law Section

Vol. 18 No. 2, Fall, 2009



• Mobile Entertainment

- Recording Contracts in a 360-World
- 20 Questions for your New Band Client
- The Music Publishing Box and How to Think Outside It
- Recent Court Decisions, Recent Legislation and Related Issues
- Protecting the Artist in Record Company Bankruptcy Proceedings

IN THIS ISSUE:

- 3. Truth, Justice, and the Canadian Way? 4
- 4. 19th Annual Entertainment Law Institute ... 13
- 7. Recent Sports & Entertainment Law Publications 22
- 8. Membership Application 24

Thank you for being a member of the State Bar of Texas Entertainment & Sports Law Section, also known as TESLAW! The 2009-2010 State Bar year is in full swing and the new TESLAW leadership plans to build on the successes of the past and work hard to increase the benefits you receive in exchange for your membership dues. As a member of TESLAW, you are currently entitled to: 1) receive the acclaimed Texas Entertainment and Sports Law Journal; 2) join the TESLAW listserve; 3) earn free CLE credits; 4) receive a discount on the cost of the annual Entertainment Law Institute; and 5) become part of the growing Texasbased entertainment and sports lawyer community. In the year ahead, the TESLAW leadership will strive to make the www.teslaw.org website the first place for TESLAW members and out of state attorneys to visit to retrieve Texas, national, and international entertainment and sports lawyer resources. Also, the official TESLAW MySpace is the social e-hangout arm of the section for Texas entertainment and sports lawyers (http:// www.myspace.com/teslaw). Additionally, the TESLAW listserve acts as the primary source of communication among the TESLAW members and between the members and the TESLAW leadership (eandslawsection@ vahoogroups.com).

So, you ask, what has occurred since the last Chairman's Report? TESLAW held its Annual Meeting and CLE in Dallas on June 26, 2009. The Section elected the new officers to serve during the 2009-2010 fiscal year. The current Council members and officers are identified on the front cover of this journal. We were fortunate to secure top speakers for the CLE presentation who presented three interesting and informative sessions covering music and sports. Dr. E. Michael Harrington, a nationally recognized expert in copyright infringement matters, gave a presentation entitled Music Copyright Protection in Business. Ted Goldthorpe at Sony AV music publishing spoke passionately on The Past, Present and Future of Music Publishing. Both presenters used music clips to inform their talks (and to get the standing room only crowd dancing in their seats). Steve Johnston, General Counsel for the Oakland A's, gave a humorous and fascinating look at the behind-the-scenes life of a general counsel in the glamorous world of major league baseball in a presentation entitled "Current Issues in Sports Law."

The demand for the "Rock Star Attorney" t-shirts continues. For any, repeat any, Texas entertainment lawyer the t-shirt is a must have and the shirt has been selling well to non-entertainment attorneys as well. because non-entertainment lawyers want them too. Buy one today for your "Rock Star" attorney family members and friends. Contact me if you are interested.

What's next? The 19th Annual Entertainment Law Institute (ELI) will be held on October 1-2, 2009 at the Hyatt Regency Hotel in Austin. Mike Tolleson, the ELI's Director, has once again created a premier event for Texas entertainment lawyers. Although detailed information is found in this journal, highlights include: a roundup of recent decisions and legislation in the entertainment area, presentations on the right of publicity, "out-of-the-box" music publishing and the effect of record company bankruptcy on artists, as well as the presentation of the Texas Star Award to Shannon Jones, Jr., Founding Partner, Passman & Jones, Dallas. In addition to great presentations and CLE credit, ELI presents a great networking opportunity and a chance to get your practice questions answered. You simply cannot miss this ELI!

19TH ANNUAL ENTERTAINMENT LAW INSTITUTE

OCTOBER 1-2, 2009 ~ HYATT REGENCY HOTEL ~ AUSTIN, TEXAS

HIGHLIGHTS

- Movie Entertainment
 Recording Contracts in a 360-World
 20 Questions for your New Band Client
 The Music Publishing Box and How to Think Outside It
 Recent Court Decisions & Recent Legislation
 - Protecting the Artist in Record Company Bankruptcy Proceedings

Check out the Section's Website!

Check it out at <u>www.teslaw.org</u>. The password for members is "*galeria3*". Should you have any comments or suggestions to improve the site please feel free to e-mail Kenneth W. Pajak at <u>ken@bannerot.com</u> or the editor at <u>srjaimelaw@comcast.net</u> ...

TEXAS ENTERTAINMENT AND SPORTS JOURNAL STAFF

Sylvester R. Jaime, Editor 1011 Highway 6 South, Ste. 216, Houston, Texas 77077 281/597-9495 FAX: 281/597-9621 E-mail: srjaimelaw@comcast.net

Professor Andrew T. Solomon, Faculty Advisor South Texas College of Law 1303 San Jacinto, Houston, Texas 77002-7000 713/646-2905 FAX: 713/646-1766 E-mail: asolomon@stcl.edu Steven Ellinger, Proofing Editor 207 Carmichael Court, League City, Texas 77573-4369 713/253-1255 FAX: 281/334-0114 E-mail: sellinger@nba.com

Craig C. Crafton, Proofing Editor Cozen O'Connor 1717 Main Street, Suite 2300, Dallas, Texas 75201 214/462-3030 E-mail: ccrafton@cozen.com

FOR THE LEGAL RECORD ...

Electronic journal? ... This is the Section's first electronic Journal. All comments should be directed to your editor and they will be passed on to the Council. Obviously budget cuts motivated the move, but some hard copies will be made available to writers, law schools and promotional efforts. ...

Technology Neutral News ... Be careful in encouraging your local or favorite recruit to attend your alma mater. North Carolina State freshman Taylor Mosely got a cease and desist letter from NC State's Michelle Lee, warning of *further action*? Seems Mosely started a Facebook group John Wall Please Come to NC State for the number 1 recruit in the country. The NC State compliance director accused Mosely of failing to follow recruiting guidelines and threatened to formally disassociate him from not just the basketball program but the entire athletic program for attempting to influence the recruits college choice. Step in the Individual Rights Defense Program at the Foundation for Individual Rights in Education. Director Adam Kissel questioned the NCAA's right to impose rules at public universities. The Foundation's position is that A student doesn't lose First Amendment rights because of a contract the university signs with the NCAA. The Facebook site was renamed Bring a National Title back to NC STATE! but prominently displayed a picture of the No. I basketball recruit. Lee questioned the NCAA's keeping up with technology and stated "I think nationally the NCAA needs to further address Facebook and how these groups play a part in recruiting. Is it realistic for us to be able to monitor them? But as the legislation stands right now, this is the position we have to take.' Erik Christianson called the group's actions *technology neutral* and simply a high tech way to try to influence recruits. The NCAA spokesman stated that the "NCAA expects institutions such as NC State, ... to reach out to the creators of such groups to educate them about the rules." In the meantime four other groups encouraged Wall to select the University of Kentucky for his likely one year of college basketball. UK compliance officer Sandy Bell would not comment as to the department's actions regarding the groups. American Civil Liberties Union's Aden Fine, questioned how the NCAA as a private organization could get the university to threaten to sanction Mosely? "The school is potentially finding themselves in a tricky situation because of the NCAA rules, but that doesn't mean public university can censor lawful speech." Said Fine. The NCAA's response through Christiansen "We [the NCAA] don't see it as a free speech issue. What we see it as is a recruiting issue. We want to be sure that we limit that level of intrusion that comes into their [the recruits] lives." Christianson further stated that "courts have upheld the NCAA's right to set recruiting rules for members." But what about the rights of the students? ...

Currently on probation the University of Oklahoma, in an effort to not get into further hot water with the NCAA, has added a rules education program and software designed to monitor compliance with NCAA rules for athletes' usage of social networking sites such as Facebook and MySpace. As part of the university's compliance report, the university created a policy warning athletes that their postings must comply with a code of conduct and can be punishable with education, counseling, suspension or expulsion and with the reduction or cancellation of financial aid. Athletes are warned as part of the policy not to post pictures that would *portray them negatively* nor post contact information that agents could use to jeopardize eligibility. "Partying, drinking, and getting wasted do not qualify as real hobbies or interests" ... the policy states. The compliance report submitted by Oklahoma, as part of the sanctions related to former coach Kelvin Sampson's recruiting violations, outlines a new software system that centralizes documents, and a program that can detect when the university's computers have been used to access improper websites. Oklahoma's general counsel notes that according to the compliance report Oklahoma's secondary violations of NCAA

rules have increased since 2002. *However, this trend should not be interpreted as negative, but rather reflects more effective and through monitoring*. Eleven total secondary violations for 2002 2004 and at least thirty for every year since. The secondary violations included the baseball, wrestling, women's basketball, softball, men's tennis, rowing and football team. The Associated Press identified some of the football violations as:

- Providing too much money to a football player to return home for the holidays because of an incorrect address;
- Coach Bob Stoops and two assistants inadvertently visiting a recruit's basketball game that had been rescheduled because of an ice storm;
- Stoops mentioning at a news conference the name of a walk on player who was technically still a prospect;
- Stoops sending an impermissible text to a recruit when he thought he was sending an email from his BlackBerry device;
- Assistant coach Jackie Shipp taking a recruit to a restaurant he thought was on campus; and The program improperly providing meals and lodging worth \$361.95 on a recruit's official visit because of a mix up regarding his legal guardian....

NOT Everyone is backing down from the NCAA. Former UCLA basketball player Ed O'Bannon, was the lead plaintiff in a lawsuit filed in San Francisco. The suit was filed on behalf of Division 1 A football and basketball players. The suit alleges that the NCAA has illegally deprived former student athletes from "myriad revenue streams." The main issue is that the plaintiffs claim that they never gave the NCAA permission to use their images and likeness after they left school in "DVDs video games, memorabilia, photographs, television rebroadcasts and use in advertising..." Co counsel Michael Hausfield said "There was no contractual or other granting of the right by a former student athlete to the NCAA or any of the universities or conferences to capitalize on the use and licensing of former student athlete's images. The NCAA is just exercising it anyway." The fight is over an estimated \$4 billion market for collegiate licensed merchandise. NCAA Spokesman Bob Williams responded by saying "... the NCAA categorically denies any infringement on former or current student athlete likeness rights." O'Bannon argues, "When you're in school you're obligated to live up to your scholarship. But once you're done, you physically, as well as your likeness, should leave the university and the NCAA." Hausfeld added that the NCAA controls the revenues by using one year term form contracts. "What it does is emphasize the illegality with the Association essentially saying by reason of these annual, limited grants of right, the Association and the universities can exercise the right to use the image of the former student athlete eternally." The NCAA has defended the practice by citing so called *scholarship* papers, or NCAA Form. 08 3a. Hausfield LLC is a world class law firm which has recovered billions in class action cases involving, *inter alia*, reparations from price fixing cartels to benefactors of slave labor, while co counsel Boies, Schiller & Flexner have also been involved in large dollars cases, such as United States v. Microsoft and Bush v. Gore.

Your comments or suggestions on the Section's website may be submitted to Kenneth W. Pajak at ken@bannerot.com and as always your comments regarding the journal may be submitted to your editor at srjaimelawkcomcast.net.

Sylvester R. Jaime Editor

TRUTH, JUSTICE, AND THE CANADIAN WAY? HOW HOLLYWOOD IS NO LONGER "MADE IN THE USA."¹

by Matthew A. Foote

Mr. Matthew Foote is an Associate with Hirschler Fleischer, P.C., in Richmond Virginia, working in real estate development and finance. Matthew holds a J.D., Magna Cum Laude from the University of Richmond School of Law, a M.A. in English from the University of California, Irvine and a B.A. in English from the University of Arizona. He also served as a Legal Intern for several organizations including the Minnesota Vikings Football, LLC; Target Stores in Glen Allen, Virginia; The Rutherford Institute in Charlottesville, Virginia; and Macaulay & Burtch, P.C. in Richmond, Virginia.

I. INTRODUCTION

"Independence Day," named after the 4th of July, is a film about an alien invasion of the United States, in which all-American symbols like the White House and the Statue of Liberty are destroyed.² "Miracle" is the story of the 1980 American hockey team, which shocked the heavily-favored Soviets at the Olympics in Lake Placid, New York.³ Both films stir patriotic emotions, champion American spirit and the American dream, and seem to be as all-American as apple pie. Thus, it is difficult to imagine that such quintessentially American films, set in the United States, were actually filmed abroad.⁴ Similarly, 2003 Best Picture Academy Award winner "Chicago," which celebrates all-American jazz culture and the city of Chicago, was filmed in Toronto.⁵ Both the original 1978 "Superman" movie and "Superman Returns"-whose main character, ironically, defends truth, justice, and the American way-were filmed outside the U.S.6 Other very "American" films recently shot abroad include "Cinderella Man," "Hollywoodland," "New York Minute," "Cold Mountain," "United 93," "Brokeback Mountain," and "Blues Brothers 2000."7 However, this trend toward "runaway" film production does more than simply raise eyebrows.8 Indeed, the exodus of U.S. productions to foreign countries transfers billions of dollars and tens of thousands of jobs to foreign economies each year, and those numbers are steadily growing.9

Though not intended to be an exhaustive study of the runaway production problem, this article examines runaway film production and its ramifications. Part II discusses the different types of runaway productions, and some of their most common runaway locations. Part III examines the economic, employment, and cultural consequences of runaway productions, and Part IV analyzes the origins and causes of the problem. Part V discusses some of the more successful action taken by foreign governments to lure motion picture productions, as well as examples of some efforts by U.S. entities to curb the flow of runaway productions. Finally, Part VI suggests a possible solution to the problem based on the most successful of those examples.

II. WHAT ARE "RUNAWAYS," AND TO WHERE ARE THEY RUNNING?

Runaway productions are feature films, made-for-television movies, and television series intended for U.S. audiences but produced outside of the United States.¹⁰ Dr. Martha Jones, of the California State Legislature's Research Bureau, defined runaway productions as "films that were conceptually developed in the United States, but filmed somewhere else. If the conversation is at the federal level, runaway production goes to other countries. If at the state level, production that goes to states [other than California] is a runaway."¹¹ There are three general species of runaway productions: artificial economic runaways, natural economic runaways, and artistic runaways.¹² Artificial economic runaways are productions lured abroad by legislatively-created (i.e. "artificial") incentives.¹³ Natural economic runaways are filmed abroad because of naturallyoccurring economic conditions, like cheap labor and rent.¹⁴ Artistic runaways are productions filmed abroad for artistic reasons, like the necessity to film a story about the Roman Coliseum in Rome.¹⁵

A. DOMESTIC RUNAWAYS

As mentioned, Hollywood entertainment executives classify productions shot in states other than California as "domestic runaways."¹⁶ This paper focuses on the international implications of runaways, and therefore domestic runaways are not emphasized. However, some of the successful efforts by U.S. locales serve as useful illustrations for how to strategically stem the flow of international runaways. For example, Louisiana and New Mexico have launched unique and successful campaigns to attract productions.¹⁷

B. INTERNATIONAL RUNAWAYS

Obviously, international runaways are far more damaging from a national perspective than domestic runaways, because they deprive the United States of the rich economic and cultural rewards that entertainment productions offer. Among the foreign countries most successful in luring U.S. entertainment productions are Canada, the United Kingdom, Australia, and New Zealand.¹⁸ Morocco and Romania are also increasingly attracting productions.¹⁹

Multiple studies have shown that the shift in production of feature motion pictures away from the United States continues to grow. In 1999, the Screen Actors Guild ("SAG") and the Directors Guild of America ("DGA") commissioned the Monitor Company, a leading management consulting company, to study the problem of runaway production in the years 1990 to 1998 ("Monitor Report").²⁰ According to the Monitor Report, 27% of U.S. developed film and television productions in 1998 were runaways.²¹ That percentage had doubled since 1990, according to the report.²² The total number of international runaways nearly tripled in that time period, increasing from 100 in 1990 to 285 in 1998.²³ Made-for-television movies have been hit the hardest by runaway production. In 1998, 45% of U.S. developed made-for-television movies were runaways.²⁴ The upward trend in international runaways has continued since the Monitor Report. The Center for Entertainment Industry Data conducted a

Continued from Page 4

similar study of the years 1998 to 2005 ("2005 CEIDR Report").²⁵ Perhaps most disturbing is the finding by the 2005 CEIDR Report that the U.S. market share in worldwide production dollars spent on theatrical releases fell from 71% to 47% from 1998 to 2005.²⁶ This constitutes an average drop in production dollars of 3% per year. While the economic consequences of this drop are already significant, they could be severe if this trend continues.

1. Canada: the Most Common Destination for Runaways

According to the Monitor Report, Canada has captured the vast majority of runaways.²⁷ In fact, so many productions occur in Canada, entertainment industry people call the country "Hollywood North."²⁸ The Monitor Report revealed that 81% of the total international runaways were filmed in Canada, including 90% of the made-for-television movies.²⁹ During the time period examined, U.S. domestic feature film production grew an average of 8.2% annually, while U.S. features produced in Canada grew 17.4% annually.³⁰ Even more revealing is that U.S. domestic television production grew 2.6% annually, as compared to an 18.2% growth in U.S. television production in Canada.³¹ The 2005 CEIDR Report found that Canadian feature film production grew 179% from 1998 to 2005.³²

2. United Kingdom and Elsewhere

The United Kingdom has the largest film industry in Europe due to its production incentive programs.³³ The 2005 CEIDR Report showed that feature film production in the United Kingdom and Ireland rose 66% from 1998 to 2005.³⁴ The 2005 CEIDR Report also revealed that feature film production in Australia and New Zealand rose an astounding 531% from 1998 to 2005, and an even more amazing 927% in Eastern Europe during this period.³⁵

III. WHY SHOULD WE EVEN CARE ABOUT RUNAWAYS?

A. IMPORTANCE OF THE FILM INDUSTRY

In 2001, the U.S. Department of Commerce reported ("Commerce Report") that the motion picture industry is "one of the most economically important industries in the United States."³⁶ The Commerce Report found that the film industry directly provided over 270,000 jobs, "more than the number of workers directly employed by the steel industry."³⁷ Estimates of the jobs produced by the motion picture industry, both directly and indirectly, range as high as 480,000.³⁸

Internationally, American films are a uniquely dominant American product. Indeed, Sheldon Presser of Warner Brothers told Congress in 2005 that the film industry has a "surplus balance of trade with every single country in the world . . . no other American business enterprise can make that statement."³⁹ For example, in 2002, 70% of box office revenues in the European Union were spent on U.S. films.⁴⁰

Productions not only directly benefit film-related businesses like pre- and post-production companies, but also indirectly to an "average of 300 non-film businesses" that provide necessary services.⁴¹ Rather than one large, self-contained business, film production is made up of small, specialized production businesses as well as personal-services support. The result is that films can have a dramatic impact on the local economies where they are filmed. For example, the film "Tin Cup" resulted in a total of \$641,000 to local dry-cleaning business, hardware and lumber companies, and private business location fees.⁴² Another Kevin Costner film, "Field of Dreams," resulted in 100,000 tourists to the Iowa cornfields in the seven years following its release.43 Likewise, "The Bridges of Madison County" also reportedly increased tourism in Iowa by 20% in one year.44 Production of "A Time to Kill" generated 10,000 paychecks to Mississippi residents.⁴⁵ Reportedly, the total direct and indirect economic impact of one movie filmed in Chicago, Illinois was \$33.5 million.⁴⁶ Similarly, "Superman Returns" injected \$80 million into the local Australian economy, as well as 800 new full time jobs and over 10,000 total people employed during filming.⁴⁷

B. CONCERN OVER RUNAWAYS HAS BEEN BUILDING FOR DECADES

Runaway productions are not a new phenomenon, and major Hollywood figures have been warning about the economic consequences for years. As early as the 1940's, then-SAG President Ronald Reagan met with President Truman about the problem of runaways.⁴⁸ Twenty-eight motion picture-related unions commissioned a study in 1957 that concluded that runaway production should be a cause of growing concern.⁴⁹ In 1961, Charlton Heston testified before Congress that foreign subsidies were creating the problem of international runaways.⁵⁰ Heston and SAG unsuccessfully pleaded with Congress to "fight subsidy with subsidy."⁵¹ Their warnings went unheeded, and the problem of runaways has continued to grow.

C. CONSEQUENCES: JOB LOSS AND ECONOMIC LOSS

The Monitor Report found that the number of full time jobs lost by U.S. workers due to runaways multiplied nearly four times from 1990 to 1998.⁵² In other words, the number of lost full-time jobs in the entertainment industry rose from 6,900 in 1990 to 23,500 in 1998. This constitutes a cumulative total of 125,100 jobs, an increase of 479% jobs lost to SAG members and 200% to DGA members.⁵³ The 2005 CEIDR Report estimates another 47,000 jobs have been lost since 2000.⁵⁴

The Monitor Report found that the total economic impact of international runaways rose from \$2.0 billion in 1990 to \$10.3 billion in 1998.⁵⁵ The direct production expenditures lost from the U.S. due to international economic runaways totaled \$2.8 billion in 1998, plus another \$1.9 billion in lost tax revenues.⁵⁶ This represents nearly a 600% increase from 1990.⁵⁷ During this time period, the total percentage of films and television programs developed in the U.S. but produced in foreign countries rose from 29% to 37%.⁵⁸ In addition, the 2005 CEIDR Report estimated that a total of \$23 billion in economic benefits related to motion picture production

6

Continued from page 5

had been lost to international runaways from the years 2000 to 2005 alone. $^{\rm 59}$

The economic ramifications of runaways are illustrated by the example of what "Chicago" did for Toronto's local economy. The local production budget for the film was \$14 million—money paid to Canadian film production companies.⁶⁰ Overall, the estimated additional impact on supporting Toronto businesses—local small businesses provided all goods and services necessary to film production—was \$21 million.⁶¹ Considering that this represents the economic benefits from only one production, it is clear that runaways represent a significant economic boon swinging from U.S. to foreign businesses.

D. CULTURAL CONSEQUENCES

Whether we like it or not, the film industry represents a significant cultural entity, and Hollywood is an iconic American setting. The Supreme Court of the United States has even weighed in on the importance of motion pictures, finding them to be a "significant medium for the communication of ideas. They may affect public attitudes and behavior in a variety of ways, ranging from direct espousal of a political or social doctrine to the subtle shaping of thought which characterizes all artistic expression."⁶² Former President Ronald Reagan said that as president of SAG, he "tried to emphasize how important the movies were to American Culture."⁶³ The U.S. Department of Commerce recognized that, in a sense, films stood for America itself:

America exported stories defining a system of government that could withstand open criticism and still grow stronger (*Mr*: *Smith Goes to Washington, Gentleman's Agreement*); stories demonstrating that talent and hard work could surpass birth into a social class as determinants of wealth or fame (*Rocky*); stories about one person's ability . . . to overcome persecution and prejudice (*To Kill a Mocking Bird*); stories exploring the impact of American slavery and prejudice and the struggle to transform society into one of equal rights for all (*Roots*). Many of these American films and television programs have helped promote freedom and democratic values, the same values that encouraged throngs of people throughout the world to rise up and challenge repressive governments, contributing to the end of the Cold War, the destruction of the Berlin Wall, and the events in Tiananmen Square before the crackdown.⁶⁴

As such, many people feel that Hollywood productions that run away to foreign countries rob the U.S. of significant cultural benefits. For example, James Mangold, writer and director of "Walk the Line," says that runaway productions cause American filmmakers to "lose the ability to capture part of our own culture."⁶⁵ One director of "Superman Returns" quit the production over the decision to film in Australia, complaining "when I flew to New York to scout, I became enamored with our greatest American city. It was clear to me that this was Metropolis. As a filmmaker, I felt it was inappropriate to try to capture the heart of America on another continent."⁶⁶ Similarly, Academy Award-winning cinematographer Jack Green refused to work on "Miracle." Green said "here was a film about the American Dream, and they were shooting it in Canada. It just really disturbs me."⁶⁷

III. CAUSES OF RUNAWAY PRODUCTION

A. OWNERSHIP OF MOTION PICTURE STUDIOS BY TRANS-NATIONAL MEDIA CONGLOMERATES

Unlike the golden era of Hollywood, where movie moguls like Charlie Chaplin and David O. Selznick ran the studios, today's movie studios are managed by corporate executives.⁶⁸ Trans-national conglomerates like News Corp., Sony, Viacom, America Online-Time Warner, and Disney now own the studios. Understandably, these large corporations owe their primary duties to shareholders. Thus, their interest is not in patriotism, artistic or creative integrity, or the sentimentality of Hollywood. Rather, the trans-national conglomerates are principally concerned with making a profit. Thus, the decision that results in the highest profit margin will be made, regardless of sentimentality.

Furthermore, these trans-national conglomerates are not dependent on the movie studios as the principal means by which they earn a profit. Indeed, in 2003 the movie business accounted for only 7% of Viacom's total income, 19% for Sony, 19% for News Corp., 18% for Time Warner, and 21% for Disney.⁶⁹ Ultimately, however, the fact that movie studios are owned by such conglomerates and motivated solely by profits could be a good thing for the runaway production problem. While many may lament its effect on the creative process and the alleged lack of quality films produced, this business model should lead runaway productions back to the United States if the proper financial incentives are implemented.

After all, if studios are now looking at films solely in terms of profit and their duty to shareholders, they are bound to produce their projects in the U.S. if the federal and state governments can make it cheaper for them to do so. 20th Century Fox Executive Vice-President Fred Baron confirmed as much, saying "what we're trying to do right now is fight to film in America. But in our process, we are forced to go offshore because of prices."⁷⁰ The rise in movie production in Louisiana and New Mexico gives a strong indication that domestic tax incentives could keep films in the U.S. However, the steady increase in international runaways indicates that the current incentives offered by foreign countries are more attractive.

B. LOWER LABOR COSTS IN FOREIGN COUNTRIES

Lower labor costs contribute to natural runaway productions, in that the labor conditions are not a legislatively-created incentive for runaways. Nevertheless, the dollars saved by paying lower wages can be significant. In fact, cheap labor can save producers up to 40% in production costs, even without any artificially-created government incentives.⁷¹ For example, the minimum weekly salary of an assistant director in Canada is \$2927 U.S., as compared to \$3285 in the United States.⁷² Clearly, if this 11% savings is multiplied by the large number of cast and crew, it can constitute significant savings in production costs. Other countries provide even more dramatic evidence. For example, Australian labor rates are 25 to 35% less than those in the U.S.⁷³ In Romania, labor costs are 80% cheaper

7

Continued from Page 6

than those in the U.S.⁷⁴ A driver for a film production in Hollywood can earn \$470 per day, as compared to an equivalent position in Romania earning less than \$10.⁷⁵

Indeed, despite its setting in North Carolina during the American Civil War, "Cold Mountain" "absolutely would not have gotten made" if not for "the savings that Romania offered."⁷⁶ Similarly, Oliver Stone was attracted to Morocco's cheap labor when filming 2004's "Alexander," as movie extras there work for \$1.80 an hour.⁷⁷ These stark contrasts between U.S. and foreign labor rates could indicate a major reason why film production spending rose 927% in Eastern Europe from 2001-2005.⁷⁸

C. FAVORABLE EXCHANGE RATES

Another major contributor to natural runaways appears to be favorable exchange rates in foreign countries. Canada provides a useful example of how the exchange rate can make foreign production more financially favorable to producers. Between 1990 and 1998, when the value of the U.S. dollar increased in relation to the value of the Canadian dollar, production costs in Canada were reduced up to 23%.⁷⁹

On the other hand, the Craig Report found that evidence directly linking exchange rates and runaways is "far from conclusive."⁸⁰ Since Canadian-commissioned reports conflict with U.S.-commissioned reports, it is difficult to know whose data and conclusions are correct. After all, U.S. studies are motivated by the desire to spur legislators into action against runaway production, while foreign studies hope to deter such action. Perhaps some answers to the exchange rate debate may be provided when data is available about runaway productions happening now, with the U.S. dollar recently at its all time low against the Euro, and similarly weak against other currencies.⁸¹ Presumably, if the dollar continues its downward trend, exchange rates will be less of a factor in the runaway production problem.

D. TAX REBATES/INCENTIVES

The 2005 CEIDR Report concluded that exchange rates, cheap labor costs, and other factors are mere bonus incentives for would-be runaway producers— tax incentives (i.e. "subsidies") are the real culprit.⁸² Therefore, the countries that offer the most generous tax incentives to producers are most successful at luring runaways.

1. Canadian Tax Incentives

Canadian federal and provincial tax incentives have drawn runaways by allowing producers rebates of up to 43.5 cents on every dollar spent on productions.⁸³ According to the Monitor Report, Canada has accomplished its dramatic increase in runaways because its government "has engaged in comprehensive and aggressive, long-term strategic campaign to attract US producers. This program includes government incentives and tax rebates, which, coupled with lower production costs, have made it economically attractive for producers to film in Canada."⁸⁴

In addition to direct financial assistance, Canada and its provincial governments offer tax incentives for productions that meet certain criteria. First, the production company must be a Canadian-controlled company and it must own the copyright in the film.⁸⁵

There are multiple ways that producers can fulfill this requirement. First, the producer can simply hire a Canadian production company to shoot the film. Canadian production companies have improved in quality to compete admirably with Hollywood production companies.⁸⁶ The producer can also team up with a Canadian production company and call the film a co-production, even if the U.S. producer has all the actual creative and financial control. Alternatively, the producer can easily incorporate in Canada for a nominal fee.⁸⁷ This approach allows the producer to take further advantage of the tax incentives, as the producer becomes a Canadian taxpayer upon incorporation. The additional criteria to receive the tax incentives are that the director or screenwriter and one of the leading actors must be Canadian, 75% of the production must take place in Canada; 75% of production costs must be paid to Canadians; and it must meet certain genre and content guidelines.⁸⁸ Presuming the producer can find the necessary Canadian creative talent, the rest of the criteria are easy to fulfill. After all, the cost of labor and the exchange rate have already helped lure the producer to shoot in Canada to begin with.

If the production qualifies, the producer receives a tax credit of 25% of labor costs up to 12% of the total cost of the production.⁸⁹ Generally, tax credits are more valuable than deductions, because credits are subtracted dollar for dollar from the company's total tax liability, and credits constitute guaranteed, immediate savings. Alternatively, the producer may opt for a refundable tax credit of 16% of salaries and wages paid to Canadian residents.⁹⁰ This clearly provides incentive for producers to employ a higher percentage of Canadians, thus improving the Canadian economy by increasing jobs and the revenue flow from consumer spending and income tax.⁹¹

2. United Kingdom Tax Incentives

As mentioned, the U.K. boasts the largest film industry in Europe, largely due to a 100% tax write-off for feature films and made-fortelevision productions that meet certain criteria.⁹² Qualification is fairly easy: the majority of labor for the production must be U.K. or European Union citizens, U.K. production companies must be used for production, and at least half of the technical production equipment must be supplied by U.K. companies.⁹³ As mentioned above in regard to Canada, the producer can easily meet these criteria by hiring a U.K. production company, utilizing a co-production approach, or incorporating in the U.K.

The U.K. also lures runaway productions through sale-lease back tax benefits. In a sale-lease back, a non-U.K. producer can refinance a film by selling it to a U.K. buyer whose business is distributing films. The buyer then leases the film back to the production company at a discount over ten to fifteen years.⁹⁴ The sale price of the film is usually calculated based on the total production cost minus the lease payments, which usually equate to roughly 10% of the production cost.⁹⁵ Thus, the producer has made an up-front profit of 10%.

The U.K. tax system encourages such transactions by allowing the buyer to claim 100% of the film's purchase price as tax relief against other liabilities.⁹⁶ So, the producer is incentivized by an up-front profit of 10%, and the buyer has a tax incentive as well as the right to exploit the movie for profit. To qualify for a sale-lease

Continued from Page 7

back tax benefit from the British government, a film must meet the following criteria: it must be intended for theatrical release; it must qualify as a "British Film," meaning that 70% of its production cost must be spent on filmmaking activity in the U.K.; and 70% of its labor costs must go to citizens or residents of the E.U.⁹⁷

3. Australian Tax Incentives

Australia has been able to use tax incentives to lure high-budget productions like "Mission Impossible II," "The Matrix Revolution," "The Thin Red Line," and "Star Wars Episode III—The Revenge of the Sith."⁹⁸ Australia offers a federal tax rebate of 15% of production expenditures of eligible film and television productions.⁹⁹ In order to qualify, the production company must be an Australian resident or a non-Australian company with a permanent establishment in Australia, and the production must have a budget of more than \$15 million.¹⁰⁰ There are certain additional limitations on the rebate. For example, if the production's budget is less than \$50 million, at least 70% of the total expenditures on goods, services, or property must be supplied in Australia.¹⁰¹ Again, it bears mentioning that a tax rebate, like a credit, is generally more valuable to the producer than a tax deduction.

Australia also lures runaways by creating incentives for private Australian citizens and corporations to invest in films. These alternative tax programs, called the "10B Offset" and the "10BA Offset," provide tax offsets for Australian investors who own copyrights in "national" or "cultural" films.¹⁰² Under the 10B offset, investors can write off their entire investment for two years for films, television movies, mini-series, or episodic series made substantially in Australia.¹⁰³ The 10BA Offset provides a 100% deduction for investments on productions completed and exploited within two years.¹⁰⁴ Since the deduction is higher, the 10BA Offset also has higher content criteria, including physical production and script elements.¹⁰⁵

Producers looking for economic incentives to run away to Australia also have incentives provided by the nation's states and provinces. Tax rebates and exemptions, cast and crew rebates, grants, and free or subsidized public services all make production in Australia more economically attractive.¹⁰⁶

E. PRODUCTION FACILITIES AND TECHNICAL EXPERTISE

Once foreign companies have lured runaway productions, those countries must be able to provide facilities and expertise comparable to those the production company could find in the U.S. After all, the economic incentives will mean nothing to the producers if they are unable to get their films made to their desired quality. Accordingly, the leaders in international runaways have increased both the capacity and quality of their production facilities and expertise. For example, British Columbia, Canada boasts seventy post-production facilities, sixty shooting stages, and the ability to crew and service forty projects simultaneously.¹⁰⁷ Similarly, the United Kingdom has developed a niche in special effects, and thus has attracted a large quantity of post-production on films shot in the U.S.¹⁰⁸

The Commerce Report found that the technical expertise issue has been self-perpetuating.¹⁰⁹ Runaway productions of U.S. films

have brought Hollywood's technical experts to foreign countries, where local artisans have been able to learn from their technical expertise. Thus, the foreign technical people are able to replicate the services offered by their U.S. counterparts who trained them. Oscar-nominated cinematographer Piotr Sobocinski verified this by praising his Vancouver crew: "I worked with the best crews all over the world, but here I have a really fantastic crew . . . I never hear the word 'no.' It's really the first time in my life that I'm going on the set with pleasure."¹¹⁰ As Mark Dillon writes, "Chicago's Oscar showing is FTAC's worst nightmare: it tells other producers that not only does Canada have a cheap dollar and attractive tax subsidies, but its film workers are on par with the best."¹¹¹

IV. ARE RUNAWAYS REALLY A PROBLEM, OR ARE SOME OF US JUST PLAYING "CHICKEN LITTLE"?

Some industry experts argue that runaway productions are not quite the problem others make them out to be. For example, Allen Scott, Director of the Center for Globalization and Policy Research at UCLA, claimed in 2002 that runaway production would never constitute a serious threat to Hollywood.¹¹² Scott's argument was based on the premise that Hollywood's technical superiority would always make Los Angeles the "significant core of the industry's product range."¹¹³ In addition, Jack Valenti, former Motion Picture Association of America ("MPAA") chief, claimed in 2004 that "there has been no 'outsourcing' of U.S. motion picture jobs . . . no permanent jobs have been exported."¹¹⁴

However, the reasoning of Scott and Valenti defy the evidence. While Scott's conclusions about Hollywood's technical superiority may have had more validity in 2002, other countries have closed the gap in this area. As discussed above, Australia, New Zealand, Canada, the U.K., and other nations have developed many production facilities and sound stages that compete admirably with those in Hollywood.¹¹⁵ It is also logical to assume that market forces will determine this issue. After all, foreign production facilities will not lure U.S. producers if they are unable to deliver the quality that producers expect. Considering the dramatic rise in runaway productions found by the Monitor Report and the 2005 CEIDR Report, foreign production companies are getting the job done.¹¹⁶

Valenti's conclusions are similarly counter-factual. As discussed above, the 2005 CEIDR Report and the Monitor Report both found that runaway productions have resulted in significant loss of employment opportunities.¹¹⁷ Perhaps the context of Valenti's comments explains his inaccuracy, as Valenti was bristling from criticism of the entertainment industry's alleged "outsourcing" of jobs from Congresswoman Diane Watson (D-CA).¹¹⁸ Nevertheless, the data shows that Watson's criticisms were valid.

In addition, Hollywood's own history shows that concern over runaway productions is justified. After all, early filmmakers came to the West Coast in order to avoid patent enforcement actions by Thomas Edison, who invented the moving picture.¹¹⁹ The availability of cheap labor and land was also a major factor in the development

Continued from Page 8

of Hollywood as the heart of filmmaking.¹²⁰ Essentially, Hollywood itself was built on runaway productions. Thus, Hollywood is right to concern itself with remaining competitive with other possible venues.

V. PREVIOUS ATTEMPTS TO COUNTERACT RUNAWAYS

A. SAG'S GLOBAL RULE ONE: A GRASS ROOTS EFFORT

SAG has implemented its own grass roots effort to stem the flow of international runaways by attempting to police its members. SAG's "Rule One" declares that "no member shall work as a performer or make an agreement to work as a performer for any producer who has not executed a basic minimum agreement with the guild. Members who violate this rule could be subjected to . . . fines, suspension, and expulsion."¹²¹ SAG worries that producers who hire actors to work on foreign projects without official SAG contracts can avoid paying contributions to guild pensions, health funds, and residuals.¹²² SAG estimates that from 1996 to 2000, guild funds lost \$23 million in contributions due to work by its members on non-signatory productions.¹²³ Due to these losses, SAG claims it has been forced to raise the minimum earnings threshold for members to qualify for SAG health plans.¹²⁴

1. Opposition to Rule One

In May 2002, SAG extended Rule One internationally, and it has faced opposition. First, it is questionable whether SAG even has the authority to prevent its members from working overseas, beyond its jurisdiction.¹²⁵ In addition, SAG faces the monumental task of monitoring its 120,000 members.¹²⁶ Thus far, its enforcement has been spotty, and has apparently consisted of asking its highprofile members to refuse to work on projects of all actors on the production do not have official SAG contracts. For example, reportedly Russell Crowe demanded that the child actors on "Master and Commander: the Far Side of the World" be signed to official agreements before Crowe would continue working.¹²⁷ Similarly, Arnold Schwarzenegger refused to work on "Terminator 3" unless production was moved from Canada to California.¹²⁸ Nevertheless, SAG has had a difficult time convincing lesser-known actors, who have less leverage, to demand that foreign projects adhere to SAG standards.129

In addition, SAG has also had backlash from the Alliance of Motion Pictures and Television Producers ("AMPTP"), which complains that Global Rule One violates the collective bargaining agreement between SAG and AMPTP.¹³⁰ Predictably, AMPTP would like its producer members to have as much freedom of trade as possible, but SAG argues that AMPTP has a duty to protect American SAG members working outside the U.S. This split on Global Rule One is indicative of the general resentment that arises when unions act unilaterally to control industry-wide problems. Often, the arguable benefits of such actions to one union's members will be offset by detriments to other aspects of the industry.¹³¹

In addition to backlash within the industry and the difficulty in policing its own members, SAG has predictably faced opposition

to Global Rule One from foreigners who hope to lure runaway productions. Producer guilds from New Zealand have resolved only to enter into domestic working relationships with actors, thus directly opposing Global Rule One.¹³² The Alliance of Canadian Cinema, Television, and Radio Artists ("ACTRA") has publicly supported Global Rule One by saying it will cooperate in allowing actors working in Canada to work under SAG terms.¹³³ However, ACTRA's stance may be simply one of diplomacy, and may not indicate its long-term stance on the issue. Indeed, both AMPTP and ACTRA have indicated they may pursue legal remedies in opposition to Global Rule One, believing it undermines existing agreements between other producers associations and actor unions.¹³⁴ In addition, the Canadian Film and Television Production Association ("CFTPA") has publicly opposed Global Rule One, maintaining that Canada is ACTRA's jurisdiction. CFTPA argues that Global Rule One strips ACTRA of its jurisdiction over English language films and television in Canada.¹³⁵ Canadian producers also argue that enforcement of Global Rule One will complicate production to the point of chasing away lucrative runaway productions.¹³⁶

The lesson of Global Rule One may be that carrots work better than sticks. Any time an entity takes action to dis-incentivize runaways to foreign countries, there will presumably be restraintof-trade complaints. Therefore, perhaps the more appropriate and effective action is to use positive incentives for producers to stay in the U.S., rather than penalties for runaways. In fact, in 2004 Congress enacted legislation intended to provide such incentives for American businesses to stay in the U.S.

B. U.S. FEDERAL INCENTIVES: THE AMERICAN JOBS CREATION ACT

The American Jobs Creation Act of 2004 ("2004 Act") was enacted by Congress to curb offshore production and lure it back to the United States. ¹³⁷ While not specifically enacted for the entertainment industry, the 2004 Act appears to have had some positive effect in discouraging runaways. For example, low budget made-for-television movies had been severely affected by foreign productions prior to the 2004 Act. The number of made-fortelevision movies produced in the U.S. had dropped from 182 in 1995 to 49 in 2003.¹³⁸ After passage of the 2004 Act, made-for-television movie production in the U.S. rose back up to 84 in 2005, largely because the 2004 Act benefits productions with budgets under \$15 million.¹³⁹

The 2004 Act includes an Internal Revenue Code provision that allows for a tax deduction for the full cost of qualified films or television productions in the year the costs are incurred, regardless of when the film is released.¹⁴⁰ For productions made in the U.S. with aggregate costs under \$15 million (or \$20 million in some low income areas), producers are allowed deductions for direct costs like equipment, as well as indirect costs like development costs, financing fees, administrative expenses, and depreciation of property.¹⁴¹ Another tax deduction is available under the 2004 Act for net income derived from license sales or exchanges.¹⁴² Thus, the producers can save money on licensing the film.¹⁴³ Any type of

Continued from page 9

production can qualify under the 2004 Act, including motion pictures released theatrically or straight to video, mini-series, television episodes, or made-for-television movies.¹⁴⁴

While the 2004 Act is a step in the right direction, it is clear that the legislation did not contemplate the film industry specifically. For example, 87% of films with budgets under \$10 million dollars were already filmed in the U.S. in 1999.¹⁴⁵ Thus, the 2004 Act benefits producers of low budget films that were not running away in the first place. Consequently, the 2004 Act is a "swing and miss" in terms of addressing the runaway problem.¹⁴⁶

C. STATE EFFORTS: LURING PRODUCTIONS TO U.S. LOCATIONS

The State of Louisiana may provide a viable example for successful legislation to keep productions at home. After the economic devastation of Hurricane Katrina, Louisiana legislation has allowed the film industry to help re-invigorate Louisiana's economy.¹⁴⁷ The 2005 legislation provides an exemption from state sales and use taxes for any motion production picture spending at least \$250,000 in Louisiana.¹⁴⁸ In addition, a production company spending \$1 million or more on payroll to Louisiana residents is eligible to receive a 20% employment tax credit.¹⁴⁹ Under the bill, a Louisiana taxpayer can receive a 10% tax credit on his or her investment in a production. If the taxpayer invests over \$8 million, the tax credit increases to 15% of the investment. Consequently, a producer who films in Louisiana can save money on sales and use taxes, significant dollars in payroll, and can more easily find Louisiana investors. The result has been a boon to Louisiana's post-Katrina economy: the ten movies filmed in Louisiana in 2005 had a combined budget of \$550 million and created 600 new jobs.150

Other states have recognized the economic and employment benefit to Louisiana's film production incentives. For example, New Mexico has begun offering a combination of benefits to lure productions: tax rebates of \$.25 per dollar; no-interest loans of up to \$15 million for productions filmed mostly in the state and hiring a high percentage of New Mexico residents; and salary rebates for training New Mexico residents as crew members.¹⁵¹ The result has been positive: \$54 million paid to 3,261 New Mexico residents, and \$160 million spent in the state on film production through September, 2007.¹⁵²

D. ATTEMPTS TO COMBAT FOREIGN SUBSIDIES

The Film and Television Action Committee ("FTAC") was formed in 1988 for the purpose of "recovering American film jobs."¹⁵³ FTAC claims to be supported by a great number of entertainment industry entities, including SAG.¹⁵⁴ FTAC believes that "foreign production subsidies like those offered by Canada and its imitators do not comply with U.S. trade agreements."¹⁵⁵ Therefore, FTAC hopes that the U.S. Trade Representative will investigate such subsidies and confirm they violate trade agreements. As a result, the FTAC hopes that such subsidies will be eliminated, either through negotiation or World Trade Organization mandate.

However, success through FTAC's efforts seems improbable for several reasons. First, the 2000 CEIDR Report stated that there is no

legislative prohibition against Canadian production subsidies. The U.S. Office of Management and Budget classifies the production of motion pictures and television as a Service Industry . . . there are no protections from a trading partner who chooses to subsidize . . . the film and television production industry under the current General Agreement on Trade in Services. If the production of motion pictures . . . were classified as a Manufacturing Industry, the Canadian subsidies would fall under the dispute settlement provisions of the World Trade Organization.¹⁵⁶

In addition, the American Jobs Creation Act of 2004, as well as various state incentives like those in Louisiana and New Mexico, are similar in nature to the Canadian subsidies about which the FTAC complains. Thus, FTAC's complaints may be somewhat hypocritical. Second, the FTAC is totally dependent on the U.S. Trade Representative, which may or may not ever take action, and the FTAC has no enforcement power on its own. Lastly, the DGA and MPAA are concerned that the FTAC's proposed course of action will ruffle the feathers of foreign nations like Canada, making it more difficult and costly to export U.S. film to those countries.¹⁵⁷ In essence, the action could increase costs for producers by forcing them to film in the U.S., while also increasing costs of exporting the films. While it could deter producers from going overseas, it could also deter filmmaking in general.

VI. CONCLUSION: SUGGESTED SOLUTION TO RUNAWAYS PROBLEM

As discussed above, runaway productions are detrimental to the United States both economically and culturally.¹⁵⁸ Aside from the cultural concerns, runaways cost the United States \$23 billion and 47,000 full time jobs from 2000 to 2005.¹⁵⁹ These numbers are growing, as foreign countries like Canada, England, and Australia take more and more affirmative steps to lure the economically beneficial productions to their shores. Similarly, some states within the U.S. have enacted legislation to draw the lucrative productions.

Congress clearly recognizes the general danger of outsourcing, and has attempted to limit it by passing the 2004 Act.¹⁶⁰ However, the 2004 Act does not go far enough to compete with overseas incentives, and does not specifically address the unique nature of runaway productions. Considering that Ronald Reagan and Charlton Heston warned of the dangers of runaway productions as far back as the 1940s, the time has come for Congress to take aggressive action to ensure that the film industry remains an indisputably American industry.¹⁶¹

First, it makes sense to form a Congressional committee to analyze what other countries and U.S. states have done to successfully attract productions, and subsequently make recommendations to Congress to implement competitive incentives.¹⁶² The motion picture committee should then continue to monitor the runaway production problem, as well as ongoing foreign legislation to attract runaways, and make periodic updates and recommendations to Congress.

Assuming the committee finds that further federal legislation is necessary to protect this economically vibrant industry, legislation

Continued from Page 10

should be drafted to lure producers back to the U.S. The new legislation could be dubbed "Federal Incentives to Lure Motion Pictures to the United States Act" ("FILMUSA"). FILMUSA must be comprehensive, employing multiple tactics to entice filmmakers, and aggressive enough to compete internationally. The first consideration should be that, when all other things are relatively equal, American producers generally prefer to stay in the U.S. for logistical ease and to save on travel costs. Thus, it is not necessary for financial incentives to blow away the competition. However, the legislation should provide tax credits or rebates, rather than the less beneficial tax deductions.¹⁶³ The credits should be liberal enough to keep producers in the U.S. when balanced against costs they will save by avoiding overseas travel for casts and crew members.¹⁶⁴ The credits should be staggered to increase based on the number of jobs provided to Americans, higher budgets (thus, greater economic benefit to the filming location), and how "American" the film is, in terms of content.165

FILMUSA could also include other provisions to deter runaways, including: stricter import tariffs on runaway productions;¹⁶⁶ SAG and DGA cooperation in implementing less favorable terms in standard SAG and DGA contracts for economic runaways; provisions that automatically adjust credits based on the falling or rising value of the U.S. dollar relative to foreign currencies; and tax credits for sale-leaseback buyers and for investors like those in the U.K., Louisiana, and New Mexico. Unfortunately, there is probably no legislative solution to combat the extremely cheap labor available in developing countries. However, SAG's vigorous enforcement of Global Rule One could help in that area, especially if SAG and other industry unions can organize mutually-beneficial solutions.¹⁶⁷ This comprehensive approach could help keep most would-be runaways "Made in the U.S.A."

- Tatiana Siegel, Superman Eschews Longtime Patriot Act, Hollywood Rep., June 30, 2006, available at http://www.hollywoodreporter.com/hr/search/article_display.jsp?vnu_content_id=1002764635. "Independence Day" plot summary, http://www.imdb.com/title/tt0116629/plotsummary (last visited Aug.
- 2 3, 2009). Miracle" plot summary, http://www.imdb.com/title/tt0349825/plotsummary (last visited Aug. 13,
- 3 2009)
- 4 5
- 2009).
 Claire Wright, Fugitive Production, L.A. Daily J., Sept. 12, 2007, at 6, available at http://www.ftac.net/ html/claire-art.html.
 See Mark Dillon, They Couldn't Have Done It Without Us, Playback, March 31, 2003, at 12, available at http://www.playbackmag.com/articles/magazine/20030331/edit.html.
 Geoff Boucher, Up, Up... and Away; Superman May Be Quintessentially American, But It's Cheaper to Film Him in Australia, L.A. Times, Dec. 31, 2005, at E1. 6
- Wright, supra note 4, at 6. 8
- Wright, supra note 4, at 6. Throughout this article, the term "film" will be used interchangeably with all types of entertainment productions, including motion pictures, made-for-television movies, and episodic television series. See discussion infra Part III(C). See generally, Pamela Conley Ulich & Lance Simmens, Motion Picture Production: To Run or Stay Made in the USA, 21 Loy. L.A. Ent. L.J. 357 (2001). Martha Jones, California Research Bureau, Motion Picture Production in California 2 (Mar. 2002), http:// www.library.ca.gov/cr/b/02/01/02-001.pdf. Adrian McDonald, Through the Looking Glass: Runaway Productions and "Hollywood Economics," 9 U. Pa. J. Lab. & Emp. L. 879, 900 (2007). Id 10
- 11
- 12
- 14 15 Id. Because of the conceptual difference between artistic runaways and productions running away for economic reasons, artistic runaways are not the focus of this article. Jones supra note 11, at 46. See discussion infra Part V(C). 16
- Germany, formerly a major player on the runaway scene, has recently enacted legislation hampering its film production. As a result, the German tax fund "party is over." See Schuyler Moore, Film Finance 101: Follow the Money, Hollywood Rep., June 13, 2006. 18
- John Horn, Filimakers Are Swept Away by Romania, L.A. Times, Oct. 2, 2005, at A1.; Megan K. Stack, Down, Dirty in Morocco, L.A. Times, Oct. 16, 2005, at E1. The Monitor Company, The Economic Impact of U.S. Film and Television Runaway Film Production (1999), http://www.dga.org/news/pr_runaway.pdf (last visited Aug. 13, 2009) [hereinafter Monitor Report]. 19
- 20
- Id. at 2. Id.
- 21 22 23 24 25 Id at 2-3
- Stephen Katz, The Center for Entertainment Industry Data and Research, The Global Success of Production Tax Incentives and the Migration of Feature Film Production From the U.S. to the World: Year 2005 Production Report (2006), http://www.ceidr.org/2005CEIDRReport.pdf [hereinafter 2005 CEIDR Report].
- 26 27
- Id. at 2. Monitor Report, supra note 20, at 3.

- 28 Claire Wright, Hollywood's Disappearing Act: International Trade Remedies to Bring Hollywood Home, 39 Akron L. Rev. 739, 739 (2006). Monitor Report, supra note 20, at 3.
- - Id
- 29 30 31 32 33
- 2005 CEIDR Report, supra note 25, at 3. 1 Thomas D. Selz, et. al., Entertainment Law 3d: Legal Concepts and Business Practices 3–12 (3d ed. 2006). 2005 CEIDR Report, supra note 25, at 3.
- 34 35 Id
- Office of Pub. Affairs, U.S. Dep't of Commerce, Commerce Secretary Mineta Releases Report on the Impact of the Migration of U.S. Film and Television Production 3 (2001), available at http://www.ita.doc. gov/media/migration11901.pdf [hereinafter Commerce Report]. Id. at 5
- 37 Jones, supra note 11, at 1
- Jones, supra note 11, at 1. Statement of Sheldon Presser, Senior Vice President, Warner Bros., Burbank, CA, on behalf of Time Warner, and the Entertainment Industry Coalition for Free Trade, Testimony Before the Full Committee of the House Committee on Ways and Means, April 21, 2005, http://waysandmeans.house.gov/hearings. asp?formmode=view&id=2610 (last visited Aug. 13, 2009). Mark Litwak, Runaway Home, L.A. Lawyer, May 2004, at 26. Commerce Report, supra note 36, at 22 (citing Dallas Film Commission). The report cited an unnamed film, which spent \$420,000 on car rentals, \$136,000 on the rental of a private residence, \$66,000 on cell phone use, \$50,400 on janitorial services, \$22,000 on freeway tolls, and \$6,000 on local transportation. Id. Gail Ecomprese Uservices for a file of the service of

- 42
- Gail Frommer, Hooray for . . . Toronto? Hollywood, Collective Bargaining, and Extraterritorial Union Rules in an Era of Globalization, 6 U. Pa. J. Lab. & Emp. L. 55, 63 (2003). Jon Garon, Star Wars: Film Permitting, Prior Restraint & Government's Role in the Entertainment Industry, 17 Loy. L.A. Ent. L.J. 1, 8 (1996). 43
- 17 LOY, LA, Lin, E.J., 19 (2004). Id. Visit Mississippi Homepage, http://www.visitmississippi.org/film/ (last visited Aug. 13, 2009). Jones, supra note 11, at 31 (citing Arthur Andersen LLP, Economic Impact Study for the Chicago Film, Television & Commercial Economic Development Coalition, Sept. 1997.) 44 45 46
- Boucher, supra note 6. Valerie Yaros, Remembering Ronald Reagan: The SAG Years, Screen Actors Guild Mag., Summer 2004, 47 48 at 29, available at http://www.sag.org/Content/Public/reagan2004.pdf. Ulich & Simmens, supra note 10, at 358-359.
- Id. at 360. Id. at 360.
- 49 50 51 52 53 54 55 Monitor Report, supra note 20, at 16.
- Monitor Report, supra note 20, at 16. Id. 2005 CEIDR Report, supra note 25, at 2. Monitor Report, supra note 20, at 16. The Monitor Report figures do not include artistic runaways, or wages paid to U.S. workers. Canada commissioned its own study contesting the Monitor Report's total economic loss figures. See Neil Craig Assoc., International Film & Television Production in Canada 1 (2004) [hereinafter Craig Report]. The Craig Report claimed that runaway production to Canada constituted a far smaller loss to the U.S. when considering the \$6.5 billion in broadcast licensing and viewing fees that flowed back into the U.S. from Canada from 1998 to 2003. However, those fees would have been paid regardless of where the film was produced, so it is incorrect to subtract those fees from economic loss figures due to runaways.
- Monitor Report, supra note 20, at 3.
- 56 57 58 59 60 61 62 63 64 65 66 67
- Id. Monitor Report, supra note 20, at 7. 2005 CEIDR Report, supra note 25, at 2. Dillon, supra note 5, at 12.
- Joseph Burstyn, Inc. v. Wilson, 343 U.S. 495, 501 (1952). Ronald Reagan, An American Life 119 (1990).

- Commerce Report, supra note 36, at 8. James Mangold, Director's Commentary, Walk the Line (20th Century Fox 2005).
- Boucher, supra note 6. Mary McNamara, Down-Home Directing, L.A. Times, Oct. 9, 2005, at E28. Heidi Sarah Wicker, Note, Making a Run for the Border: Should the United States Stem Runaway Film and Television Production Through Tax and Other Financial Incentives?, 35 Geo. Wash. Int'l L. Rev. 461, 162 (2000) 68 463 (2003).
- 69
- Edward Jay Epstein, The Big Picture 16 (2005). Tatiana Siegel, Unite, State Film Czars Told, Hollywood Rep., Aug. 1, 2005. Harold L. Vogel, Entertainment Industry Economics: A Guide for Financial Analysis 72–73 (2nd ed. 70 71
- 1990).
- Carl DiOrio & Dave McNary, H'wood's Runaway Train, Variety, Feb. 4–10, 2002, at 1. 2005 CEIDR Report, supra note 25, at 70. Horn, supra note 19, at 1.
- 72 73 74
- AS union control on the industry be rise. Thus, these factors are exacerbated. Id. Id. As union control on the industry becomes increasingly tighter, labor-costs for producers continue to 75
- 76
- Id. Relocation, Relocation, Relocation, Economist, July 10–16, 2004, at 77. 2005 CEIDR Report, supra note 25, at 8. Romania joined the European Union in early 2007, so minimum wage standards could mitigate this problem in that country. Nevertheless, there are countless poor nations that could provide such cheap labor. Monitor Report, supra note 50, at 4. Craig Report, supra note 55, at 12. For example, the Craig Report found that in 1999, when the Canadian dollar was at its weakest, studios spent \$977 million on Canadian production. Id. However, as the value of the Canadian dollar rose in 2002 and 2003, runaway production continued to rise as well—reaching \$117 billion in 2003. Id. This could suggest that exchange rates are simply one factor among many entered of the Canadian dollar rose in 2002 and 2003, runaway production continued to rise as well—reaching \$1.17 billion in 2003. Id. This could suggest that exchange rates are simply one factor among many entered into the cost-benefit analysis equation undertaken by studios. Despite the falling U.S. dollar, foreign tax incentives may still be strong enough to lure runaways. John W. Schoen, Should We Be Worried About the Falling Dollar?, http://www.msnbc.msn.com/ id/21018869/ (last visited Aug. 13, 2009). 2005 CEIDR Report, supra note 25, at 4. Ian Caddell, Residual Effect, Hollywood Rep., Feb. 19, 1999, at 21; David Finnigan, Woe Canada: Guilds Cite 510B in Lost Prod'n, Hollywood Rep., June 25, 1999, at 1. Seel zet al., supra note 33, at 3-10. See Paul A. Baumgarten, Donald C. Farber & Mark Fleischer, Producing, Financing and Distributing Film 154 (2d ed. 2004); Litwak, supra note 40, at 24. See discussion, infra Part III(E). John W. Cones, 43 Way\$ to Finance Your Feature Film: A Comprehensive Analysis of Film Finance 176 (1995).
- 81
- 83 84
- 85
- 86 87 (1995)
- 88 Canada Revenue Agency, supra note 85. 89
- Id
- 90 91 92 Id
- See U.K. Film Council, Filming in the U.K., http://www.ukfilmcouncil.org.uk/filminginuk (last visited Aug. 13, 2009); Commerce Report, supra note 36, at 77. Commerce Report, supra note 36, at 77. See Stuart Kemp, UK Film Biz Gets Tax Break Commitment, Hollywood Rep., Mar. 18, 2004, at 1, 22.
- 93 94 95 96 97 98 Id
- Id. See U.K. Film Council, supra note 92. Selz et. al., supra note 33, at 3–13, n. 19.

99 See Australian Film Commission, Ausfilm Factsheet: Screen Production Incentives in Australia, http://www. isfilm.com.au/images/stories/AusfilmFactsheets/incentives%20may.pdf (last visited Aug. 13, 2009). aus Id 100

Continued from page 11

- 101 102
- See Income Tax Assessment Act of 1936, http://www.ausfilm.com.au/file_banks/ausfilm/misc_pdf/
- Ausfilm_Incentives_Sticker.pdf; Litwak, supra note 40, at 24; Selz et. al., supra note 33, at 3-14. Income Tax Assessment Citation, supra note 102. 103
- 104 105 Id
- Id 106
- Lionel S. Sobel & Donald E. Biederman, International Entertainment Law 280–1 (2003); 2005 CEIDR Report, supra note 25, at 46–50. BC Film Commission, Industry Profile, http://www.bcfilmcommission.com/about_us/industry_profile.htm 107
- (last visited Aug. 13, 2009). Commerce Report, supra note 36, at 53-54. 108
- 109
- 110 Dillon, supra note 5.
- Id. Allen J. Scott, A New Map of Hollywood and the World, Center for Globalization and Policy Research, School of Public Policy and Social Research, UCLA 22–23 (2002), available at http://www.ersa.org/ ersaconfs/ersa02/cd-rom/papers/521.pdf. 112
- 113
- ersaconfs/ersa02/cd-rom/papers/521.pdt. Id. at 22. Peter Kiefer, Valenti Defends the Right of U.S. Firms to Film O'seas, Hollywood Rep., Apr. 19, 2004. Commerce Report, supra note 36, at 46-56. See discussion, supra Part III(E). In 1998, 90% of the budgets for the top 250 films were spent in the U.S. By 2001, that number had dropped to 76%. Stephen Katz, The Center for Entertainment Industry Data and Research, The Migration of Feature Film Production from the U.S. to Canada and Beyond: Year 2001 Production Report 1, available at http://www.ceidr.org/y2/k1report.pdf [hereinafter 2001 CEIDR Report]. Canada's tax incentives caused its percentage of total U.S. film budgets to rise from 10% in 1998 to 24% in 2001. Id. Astoundingly, the U.S. market share of production dollars spent on theatrical releases plummeted from 71% to 47% between 1998 and 2005. 2005 CEIDR Report, supra note 25, at 2. Clearly, the efforts of foreign countries to attract runaway productions are working. 115 116 Epstein, supra note 69, at 5. See Kiefer, supra note 114.
- Epstein, supra note 69, at 5
- 119 120 Ιđ
- Global Rule One, Frequently Asked Questions, http://www.sag.org/content/global-rule-one (last visited Aug. 13, 2009). See Dave McNary, Cannes Heat on SAG Rule 1, Daily Variety, May 20, 2002, at 5; Peter Kiefer, Rule Const Hulberged Bene Fich 27, 2002 et al. 121
- 122 One, Take One, Hollywood Rep., Feb. 27, 2002, at 1 123 Id.
- Id
- 124 125 Overseas Producers Blast SAG's Global Rule One, http://www.imdb.com/news/sb/2002-05-20 (last visited Aug. 13, 2009). Selz et. al., supra note 33, at 3-7. Id. at 3-15.
- 126 127 128
- Id. at 3-15. James Bates, Can Sacramento Help Hollywood?, L.A. Times, Oct. 20, 2003, at B1. Kiefer, supra note 112, it 1. McNary, supra note 122; Etan Vlessing, Canadians Support SAG's Global Rule One, Hollywood Rep., Apr. 18, 2002, at 1, James Bates, SAG Calls for Guild Contracts Overseas, L.A. Times, Apr. 29, 2002, at B1; Ray Bennett & Peter Pryor, SAG Implements Global Rule One, Hollywood Rep., May 7, 2002, at 59. The benefits to SAG members are questionable to begin with, as limiting their right to work overseas could indeed be detrimental to all but those actors with the most leverage. See supra note 125 and accompanying text. McNary, supra note 122; Etan Vlessing, Canadians Support SAG's Global Rule One, Hollywood Rep., Apr. 18, 2002, at 1, James Bates, SAG Calls for Guild Contracts Overseas, L.A. Times, Apr. 29, 2002, at B1; Ray Bennett & Peter Pryor, SAG Implements Global Rule One, Hollywood Rep., May 7, 2002, at 59. See Let. al., supra note 33, at 3-16, n. 33. See McNary, Vlessing, Bates, Bennett & Pryor, supra note 130. Selz et. al., supra note 33, at 3-16, n. 33. Id. American Jobs Creation Act of 2004, Pub. L. No. 108-357, 118 Stat. 1418 (2004)[hereinafter 2004 Act]. 130 131
- 132
- 134
- 135 136
- American Jobs Creation Act of 2004, Pub. L. No. 108-357, 118 Stat. 1418 (2004)[hereinafter 2004 Act]. 137

- Selz et. al., supra note 33, at 3-11, n.10.
- 139 Id Id. 26 U.S.C.A § 181 (2004); 2004 Act, supra note 137. Previously, such costs had to be amortized over a number of years. The 2004 Act allows production companies to take the full deduction in the year the cost is incurred. Thus, the studio saves money up front. See Selz et.al., supra note 33, at 3-14; Gunnar Erickson, Harris Tulchin & Mark Halloran, Independent Film Producer's Survival Guide 113-115 (3d ed. 2005). 140
- That is full in a water fail of an independent film from the solution of the first form of the cost of the solution of the cost of the solution of the cost of the solution of the cost o 141 142 143
- 144 145
- 147 148
- The tax deduction increases by year, phasing in at 3% in 2005–06, 6% in 2007–09, and 9% thereafter. See 26 U.S.C.A. 199(B)[10a)(1). Productions that "depict actual sexually explicit conduct" as defined by 18 U.S.C.A. § 2257 are not included. See 26 U.S.C.A. § 8 [181, 191. 2001 CEIDR Report, supra note 116, at 3. . Moore, supra note 18. Missy Schwartz, Southern Comfort, Ent. Wkly., March 24, 2006, available at http://www.ew.com/ew/ article/0,.1174193,00.html. H.B. 731, 2005 Reg. Sess. (La. 2005). Id. Again, a tax credit comes dollar for dollar off of the company's tax liability—so it generally represents a greater savings than does a deduction. Schwartz, supra note 147. 149
- 150 Schwartz, supra note 147.
- 2005 CEIDR Report, supra note 25, at 65. New Mexico State Investment Council, Film Investment Performance Summary, Sept. 30, 2007, at 5, available at http://www.sic.state.nm.us/PDF%20files/NM_Film_Investment_Program_93007_Final_ Revision.pdf.
- Film and Television Action Committee, About FTAC, www.ftac.net/html/about.html (last visited Nov. 26, 2007)[hereinafter FTAC]. 153
- 154 155 Id. Id.
- Stephen Katz, The Center for Entertainment Industry Data and Research, The Migration of Feature Film Production from the U.S. to Canada: Year 2000 Production Report 5 (2001)[hereinafter 2000 CEIDR 156 Report].
- 157 FTAC, supra note 153
- 158 159
- See discussion supra Part III. 2005 CEIDR Report, supra note 25, and accompanying text. 2004 Act, supra note 137.
- 160
- 162
- 163
- 2005 CELDR Report, supra note 25, and accompanying text.
 2004 Act, supra note 137.
 Yaros, supra note 48; Ulich & Simmens, supra note 10, at 360.
 The United States is the only major nation without an organization at the federal level to address the motion picture industry. Commerce Report, supra note 36, at 89.
 This is the "subsidies to fight subsidies" approach encouraged by the DGA and MPAA. Ulich & Simmens, supra note 10, at 360.
 Foreign film companies lured to the U.S. by FILMUSA and the falling dollar could also benefit from the tax incentives to the extent they have taxable earnings in the U.S. Therefore, there should be no concern from the World Trade Organization over discrimination of foreign companies filming in the U.S. Indeed, FILMUSA would not differ dramatically from Cnandian subsidies, which have already been ruled in compliance with the Trade in Services Agreement. See 2000 CEIDR Report, supra note 156; see FTAC, supra note 153; see generally General Agreement on Trade in Services, Apr. 15, 1994, Marrakesh Agreement Establishing the World Trade Organization, Annex II B, Legal Texts: Results of the Unguay Round of Multilateral Trade Negotiations 284 (1999), 1869 U.N.T.S. 183, 33 I.LM. 1167 (1994)[hreeinafter GATS].
 Films with quintessential American themes, like "Superman," 'Chicago," and "United 93," would receive greater incentives for filming in the U.S. While this suggestion may raise fair-trade concerns, there are similar provisions in Canada, England, and Australia that have y et to incur WTO disapproval. See Wright, supra note 14 (arguing that, unlike Canadian subsidies, U.S. subsidies are not "trade-distorting," and thus would not invite WTO action).
 Subject to compliance with the WTO's Trade In Services Agreement. See GATS, supra note 164; but see
- Subject to compliance with the WTO's Trade In Services Agreement. See GATS, supra note 164; but see Wright, supra note 28, at 745 (arguing "the U.S. Government could [under WTO law] also initiate an action to impose countervailing duties on the subsidized films when they are imported into the U.S."). 166
- 167 See supra note 131 and accompanying text.



with Gordon P. Firemark & Tamera H. Bennett

Entertainment Industry News, Information and Discussion for lawyers and industry professionals.

http://entertainmentlawupdate.com



*in some states, pending approval elsewhere.

TexasBarCLE presents the 19th Annual

Entertainment Law Institute

Cosponsored by the Entertainment Law Section of the State Bar of Texas

LIVE Austin October 1-2, 2009 Hyatt Regency Hotel 10.30 Break **Review of Recent Legislation and** Thursday 3.15 Related Issues 1 hr 7.5 hours including 1 hour ethics 10:45 The Texas Music Industry Today Moderator Stan Soocher, Denver, CO 5 hr 7:45 **Registration and Continental** Casey Monahan, Austin Editor-in-Chief, Entertainment Law & Breakfast Director, Texas Music Office Finance and Associate Professor, Music & Entertainment Industry Studies 8:15 Welcoming Remarks 11:15 Texas Film Industry Update .5 hr University of Colorado Denver **Course Director** Bob Hudgins, Austin Jay Rosenthal, Washington, DC Mike Tolleson, Austin Director, Texas Film Commission Senior Vice-President & General Mike Tolleson & Associates Counsel 11:45 Break – Lunch Provided National Music Publishers' Association 8:30 **Roundup of Recent Court** Decisions 1 hr 12:05 **Texas Star Award Presentation** John L. Simson, Washington, DC Stan Soocher, Denver, CO **Posthumously Honoring** Executive Director, SoundExchange Editor-in-Chief, Entertainment Law & Shannon Jones, Jr. (1927-2009) Finance and Associate Professor, Music Founding Partner, Passman & Jones, The Music Publishing Box and How 4.15 & Entertainment Industry Studies Dallas to Think Outside It 1 hr University of Colorado Denver The music publishing industry is Jerry C. Alexander, Dallas undergoing a transformation, trying to 9:30 Copyright 101 1 hr Receiving the Award for Passman & squeeze new revenue sources into the Copyright basics, the mechanics of Jones existing business models. A discussion registration using the new on-line of traditional income sources and the systems, when copyright protection 12:15 Right of Publicity in Texas and new music publishing economy. attaches, term of copyright, the need Beyond .5 hr Ron Sobel, Studio City, CA for registration before litigation and Maria Elena Holly, Dallas Partner, Winogradsky/Sobel how registration impacts the ability to Estate of Buddy Holly President, North Star Media collect both statutory damages and Jonathan Faber, Indianapolis, IN attorney fees. Steve Winogradsky, Studio City, CA Attorney/Luminary Group LLC Partner, Winogradsky/Sobel Amy E. Mitchell, Austin Attorney and Mediator at Law 12:45 Break 5:15 Adjourn Steve Winogradsky, Studio City, CA 1:00 **Malpractice and Ethics Update** Partner, Winogradsky/Sobel Friday 1 hr ethics Coyt Randal 'Randy' Johnston, 6 hours including .5 hour ethics Dallas Johnston•Tobey **Continental Breakfast MCLE CREDIT** 8.00 "I Won't Back Down"- Protecting 2:00 **13.5 HOURS (1.5 ETHICS)** 8:20 Announcements the Artist in Record Company MCLE COURSE NO: 901180663 Bankruptcy Proceedings 1 hr Applies to the College of the State 8:30 20 Questions for your New Band What to do when your artist Bar of Texas. Client 1 hr (.25 ethics) client's record label or label client's The importance of band agreements distributor files for bankruptcy. **CPE CREDIT** and the issues they should address. J. Craig Barker, Austin TexasBarCLE is registered with the Moderator Law Office of Craig Barker Texas Board of Public Accountancy Amy E. Mitchell, Austin to offer courses. The State Bar's James V. Hoeffner, Austin Attorney & Mediator at Law continuing education sponsor I.D. Graves Dougherty Hearon & Moody Layne Lauritzen, Austin number is 135. Since CPE credit is Layne Lauritzen, CPA, PC calculated on a 50-minute hour, this 3:00 Break course totals 16.25 hours.

TexasBarCLE Advanced Courses are sponsored by LexisNexis, a division of Reed Elsevier Inc.

Lisa Whynot, *Toronto, Canada and Los Angeles, CA* Attorney & Counselor at Law

9:30 Recording Contracts in a 360 World 1 hr A mock negotiation of a 360-style

recording agreement. Dina LaPolt, *W. Hollywood, CA* LaPolt Law

Doug Mark, Los Angeles, CA Mark Music & Media Law

10:30 Break

10:45 Looking Behind the Curtains in an Entertainment Industry Dispute: A View from the Mediator's Chair 1 hr (.25 ethics) An examination of the dynamics that occur in resolving entertainment related disputes by lawyers who are in the trenches.

> <u>Moderator</u> D'Lesli M. Davis, *Dallas* Fulbright & Jaworski

Greg Bougeois, *Austin* Lakeside Mediation Center

Eric Galton, *Austin* Lakeside Mediation Center

Jeff Krivis, *Encino, CA* First Mediation Corporation

11:45 Lunch on Your Own

1:00 Digital Distribution of Audio-

Visual Content 1 hr Content providers, aggregators, distributors, and exhibitors, all play a role in the path content takes from the creator to the end user. Also discussed will be common licensing issues affecting the process.

<u>Moderator</u> Jed Lackman, *Universal City, CA* Vice President, Business & Legal Affairs Universal Studios Home Entertainment

Matt Dentler, *New York, NY* Cinetic Media, Inc.

Speaker to be Announced

2:00 Break

2:15 **Mobile Entertainment** *1 hr* Mobile media technology is a constantly changing landscape. Representatives from various segments of the industry—content providers, distributors, and attorneys discuss key tech concepts, how new media issues inform all aspects of entertainment practice, and emerging trends.

> <u>Moderator</u> Steve Masur, *New York, NY* Attorney

Dave Dederer, *Seattle, WA* Vice President, Business Development Melodeo, Inc.

Speaker to be Announced

3:15 Entertainment Property Financing 1 hr Angels, venture capitalists, and private investors are out there. Discover what the deals look like and how to avoid tripping over securities regulations.

> <u>Moderator</u> Deena Kalai, *Austin* Deena Kalai, PLLC

Nicholas J. Fox, *Austin* Vinson & Elkins

Sally Helppie, *Dallas* Tipton Jones Law Sabbatical Pictures

4:15 Adjourn

For their generous support of this program, TexasBarCLE and the Entertainment and Sports Law Section of the State Bar of Texas would like to recognize and thank:





2009 TEXAS STAR AWARD RECIPIENT

Shannon Jones, Jr. April 27, 1927 – February 12, 2009



S hannon grew up in Kaufman Texas, where he first heard and learned to love and then play and sing Country and Western music. Upon his honorable discharge from the service in World War II, Shannon returned to Texas, obtaining an undergraduate degree from Texas A&M University and law degree from Southern Methodist University. In 1950, he began practicing law in Dallas with the law firm of

Alexander, George, Thuss & Passman and on January 1, 1952, founded Passman & Jones with Sam Passman.

Shannon's law career spanned over fifty years in the practice of civil law and in his later years was concentrated primarily in the fields of wills and estates and entertainment. He played a vital role in counseling many business leaders as well as musicians, including "ghostwriting" for many artists primarily in the Country and Western field, such as Bob Wills and other successful songwriters. Shannon was also a champion of artist's copyrights and fair compensation for their works. He was a pioneer in the industry who drafted the Texas right of publicity law and along with Maria Elena Holly lobbied for and secured its passage which now protects the name and likeness of artists like Buddy Holly for fifty years after their death. Shannon also founded the first Entertainment Law and Study Group at the Dallas Bar Association. fyi

INFORMATION

Ways to Save on This Course!

Register by September 17 and Save \$50

because early registrations help us ensure that sufficient course books, seating, and refreshments are available. No confirmation will be sent. Course materials and seating will be reserved for pre-registrants until the start of the first topic.

Entertainment and Sports Law Section Members Can Save \$75

by registering by September 17, 2009 (thereafter, save \$25); see the registration form. Not a member? To join, contact the State Bar's Sections Coordinator, Kathy Casarez, at 800-204-2222, x1420. Please do not send Section membership payment with your registration fee.

Earn Discounts by Registering Five or More

from your firm or agency. Contact Firm and Group Sales Representative Laura Angle of TexasBarSolutions at 512-250-5575 or langle@texasbar.com.

State Bar Policy Manual §7.03.06 Allows Some Judges to Attend Free

provided (1) the course is directly related to their tribunal's jurisdiction, (2) they are full-time judges or judges retired under the Texas Judicial Retirement System and (3) space is available for all paying registrants. This privilege does not extend to receivers, trustees, court staff, or persons serving part-time in any judicial capacity. NOTE: Judges must actually attend to receive course materials.

Parking Information

is subject to change without notice and is not a guarantee of available space. See "Parking" to the right for details.

If You Need Special Accommodations to Attend,

please contact us as soon as possible at 800-204-2222, x1797.

Registered But Can't Attend? Still Earn Participatory Credit.

After the course you'll automatically be sent the written materials and may view the course online later; see "FREE TO OUR REGISTRANTS!" below. However, if you wish a refund instead, mail or fax (512-427-4111) your request so that we receive it at least one business day before the program.

FREE TO OUR REGISTRANTS!

Online Videos of the Presentations and PDFs of the Course Materials will be available to registrants 6-8 weeks after the course. Provide us your email address when registering and we'll alert you when these benefits are available and how to access them. (Note: Presentation lengths may vary from that which was advertised.) There's no extra charge; as a registrant, you're already entitled!

Policy of Impartiality

The State Bar of Texas does not endorse political candidates. When a candidate for public office is included in promotion for or participates in a TexasBarCLE event, the State Bar is not taking a position for or against anyone's candidacy.

Reserve Your Hotel Room Now!

A block of rooms at a special rate is available on a first-come, first-served basis.* Inform the Hyatt that you will be attending this State Bar of Texas Course.

AUSTIN Live

Hyatt Regency Austin 208 Barton Springs Road (512) 477-1234 \$185 for a single/double



Deadline:

August 31, 2009

* Because the Austin City Limits Music Festival begins on October 2, this special room block could fill well before the August 31 deadline. Don't delay, reserve your room now!

Hotel Parking

- Valet parking: \$18/day (unlimited access)
- Self-parking: per hour basis up to \$12/day

Transportation

• Taxi: About \$20-\$27 one-way.

• Airport Super Shuttle (512) 258-3826: \$12 oneway. From the airport to the hotel, no reservations are needed. From the hotel to the airport, you must arrange for pickup in advance via the concierge at the hotel. Transportation is available 24 hours/day.

Directions to the Hyatt from Austin Bergstrom International Airport

Exit airport on Presidential Blvd. Turn left on to TX-71. Turn right on to E. Riverside Dr. Turn right on to Barton Springs Rd. Hotel is on the left.

Our thanks to our sponsor State Bar of Texas Insurance Trust



State Bar of Texas

Roland Johnson	President
David Fisher	Chair of the Board
Terry Tottenham	President-Elect
Harper Estes	Immediate Past President
Cori Harbour	
F. R. "Buck" Files, Jr	Chair, CLE Committee
John Sloan	. Chair, Board PDP Committee
Michelle E. Hunter	Executive Director
Patrick A. Nester	Director, TexasBarCLE

Course Director Mike Tolleson.....Austin

Planning Committee

J. Craig Barker	Austin
Tamera H. Bennett	Lewisville
D'Lesli M. Davis	Dallas
Edward Z. Fair	Austin
Deena B. Kalai	Austin
Amy E. Mitchell	Austin
Stan Soocher	Denver, CO
Alan W. Tompkins	Dallas
Donald J. Valdez	Fort Worth
Steve Winogradsky	Studio City, CA

TexasBarCLE Planning Team Kandi Botello.....Program Coordinator Montgomery Miller.....Sr. Program Attorney

Entertainment and Sports Law of the State Bar of Texa	
D'Lesli M. Davis	Chair
Don Valdez	Chair-Elect
Mitzi Brown	Secretary
Shannon Jamison	Treasurer
Alan W. TompkinsImmediat	e Past Chair

E-Mail:_

Registration Form	WAYS TO REGISTER
Online TexasBarCLE.com credit card onlyPhone 800-204-2222, x1574 during business hours credit card onlyFax 512-427-4111 credit card only	Mail State Bar of Texas - CLE LB #972298 P. O. Box 972298 Dallas, TX 75397-2298 check or credit card Registering five or more? Email Laura Angle at laura.angle@texasbar.com for group discount information.
Save up to \$75! Entertainment Law Ir	nstitute 2009 8095
 Entertainment Law Institute Austin LIVE, October 1-2\$445 Take \$50 off: Register by September 17 Registration fee includes breakfast both days and a special luncheon presentation on Thursday. 	 2 I want the format of my course materials to be: a Electronic (PDFs on a USB drive) b Hardcopy (1 notebook) b BOTH so I'll add \$80
 I am entitled to 1 or 2 discounts: \$50 OFF because I'm registering by September 17, 2009. (If by mail, date of postmark will be determinative.) \$25 OFF because I'm a current member of: Entertainment & Sports Law Section, State Bar College, or the State Bar Legal Administrators or Legal Assistants Division -OR- I'm licensed 2 years or less. \$ is my total for the course and materials. As a judge, I AFFIRM I qualify under \$7.03.06 (State Bar Policy Manual) for complimentary admission (see pg 4.) 	 I can't attend. Just send course materials.* Course Book
J am paying by: Check (enclosed) payable to the State Bar of Texas for \$	charge (see "Free to Our Registrants" on p. 4). [†] Plus 8.25% tax on total. Please include tax or attach exemption certificate. Orders for books, USBs, and audio CDs are filled 4-6 weeks after the live event and shipped separately. A bill will be sent unless a credit card charge is authorized on this form.
Address for Bar-Related Mail Zip City/State Zip Phone () Fax ()	• 800-204-2222 x1574 • 512-427-1463 x1574

Legal BITS!

BAD GUYS OR GOOD GUYS? You be the JUDGE!

Texting? Stadiums which have the crowds watching each other? At issue is technology! Able to keep the stands calm and the venues safe, security personnel are ever on the alert. Stadium command posts are being set up to accept test messages from fans! If the guy next to you is suspicious looking or maybe too loud? Text the complaint to police personnel able to respond in an instant. Text messaging offers *efficient and discreet* reporting of trouble to security personnel. Safety or an infringement on liberty?



NCAA did not identify the ineligible player, but Derrick Rose, current NBA point guard, was reported to be the guy. Rose was accused of having someone else take the SAT for him. The deception apparently worked and Rose was eligible as a freshman after failing the ACT 3 times. The school defended their investigation and permitted the player to participate. But the SAT team did their own investigation and told the player, the university and the NCAA the test the player took had been canceled. Of course, by then Rose had

already been the No. 1 pick in the 2008 draft of the Chicago Bulls. After all the steps to protect the test, still a way to beat it.

Bad Guys or Good Guys?

Charmon Sinkfield, arrested and charged in the robbery and shooting death of former boxing champ Vernon Forest. Jquante Crews, the alleged get away driver, was charged with murder, possession of a firearm by a felon and armed robbery. After all the blows the champ took in the ring, Atlanta police catch a shooter and a driver who would have never met Forest in a fair fight in the ring.

Bad Guys or Good Guys?

Jeremy Tyler, jumped from high school to \$140,000.00 for playing pro basketball. Tyler skipped his senior and college ball to sign with Israeli Premier League team Maccabi Haifa. The 6' 11", 260 pound, high schooler, averaged 28.7 in his junior year, and decided the money was better than his senior year. After being bored with high school, the kid goes overseas for the money.

Bad Guys or Good Guys?

Rick Pitino, accused of rape by Karen Sypher, after she was busted for blackmail when she attempted to extort millions from the Louisville coach. After confessing to the affair and admitting the mistake, Pitino faces pressure from the media, while he attempts to comfort his wife and coach the nationally ranked Cardinals, while fans, friends and the university point to his coaching record as the reasons to support the coach. After being caught, Pitino has taken the offensive and pointed at Sypher as a liar, *bat crazy* and the *alleged criminal*.

Bad Guys or Good Guys?

Julio Castillo made a pitch to the wrong fan. Castillo was convicted of felonious assault for throwing a baseball into the stands. The ball injured a fan during a 10 minute melee. Judge Connie Price, of Montgomery County Ohio Common Pleas Court, gave the pitcher

30 days in jail and a pitch that if he violated any probation conditions he faced 3 years in prison. After the conviction and concussion to the fan, a just sentence or going easy on a ball player?

Bad Guys or Good Guys?



Bad Guys or Good Guys?

Football turning to fun, fair, positive players? The American Football Coaches Association is encouraging teams to change the major college football landscape, asking for a before the game pre kickoff handshake. Softer and friendly players or just another infringement on the game?

Bad Guys or Good Guys?

Tim Donaghy back in the federal can. Probation violation: NOT SHOWING UP FOR WORK! At 42, the former NBA referee faces more time on the gambling charges. After he was accused of taking thousands of dollars for inside tips on NBA games, he pleaded guilty to conspiracy and wire fraud in interstate commerce. Already at a half house, Donaghy instead is now back behind federal bars for not understanding the federal rules during his release.

Bad Guys or Good Guys?

Ivo Minar, tested positive after a Davis Cup match. The Czech Tennis Association said Minar was tested after a quarterfinal win in Argentina. The Association reported that an unspecified supplement of a banned stimulant was found in Minar's test. Minar pulled out of the U. S. Open saying he had an injury. After the Association, before the match, told him he would be tested?

Bad Guys or Good Guys?

National Football League former player Plaxico Buress pleaded guilty. Indicted on 2 counts of criminal possession of a weapon and one count of reckless endangerment, he faced a 3 1/2 year minimum prison term if he had been convicted. The former New York Giants receiver accepted a plea bargain with a two year prison sentence for accidentally shooting himself in a New York bar. After forcing fans to hear *not guilty*.

Bad Guys or Good Guys?

Memphis, John Calipari, fans, record books all get big changes from the NCAA. Stripped of all wins from 2007 08, for using an ineligible player, the university is not going down without a fight. Shirley Raines, Memphis president, is leading the appeal for an *unfair penalty*. The Tigers lost the national champion in 2007 08 to the University of Kansas. Coach Calipari also had other teams stripped of appearances, the 1985 Memphis and Massachusetts teams. The

Congratulations

Austin City Limits. After 35 years of *iconic and emerging, cutting edge artists* and over 800 performances from assorted musical genres, the show received the official historic designation by the Rock and Roll Hall of Fame and Museum. Providing inspiration to the Austin music scene by staging the likes of Johnny Cash, Willie

Nelson, and Bonnie Ratt, Austin City Limits start its 35th season on Oct. 3. Executive producer Terry Lickona hailed the designation for the venue which has hosted performers from around the world. Museum president Terry Stewart bestowed the honor on the program citing its *incredible archive of material*.



Willie Nelson sang in Austin City Limits' pilot in 1975 (above left) and fronted the country swing band Asleep at the Wheel more recently (above right). He's also partner in the big Austin redevelopment that will house the program's new theater.





Heard at the Supreme Court

The Supreme Court will consider a case potentially significantly changing antitrust law. Spokesman Greg Aiello called the NFL ... *the unique structure of a sports league*. The NFL has taken the position that the Court should extend nationally the favorable appeals court rulings. The NFL won the federal appeals court antitrust case in Chicago, but elected to

appeal to the Supreme Court in an effort to address frivolous antitrust lawsuits. American Needle Inc, of Buffalo Grove, Ill., filed an antitrust lawsuit challenging the NFL's deal with Reebok. American Needle was a company that provided headwear to the NFL. The NFL in 2001, from among several sports merchandisers, entered into an exclusive deal with Reebok, which American Needle said violated antitrust law. The question to be considered: *Is the league a single entity* that can act collectively, as the NFL, or 32 distinct businesses subject to antitrust laws? Law professor and director of the National Sorts Law Institute at Marquette University (and former journal advisor), Matt Mitten was quoted as saying: "This will be the first time the Supreme Court will consider the merits of the single entity defense." He added, "A favorable court decision could give the league a lot more room not to have to fear suits on issues such as relocation and ownership."

Gabe Feldman, associate law professor and director of the Sorts Law Program at Tulane University, said "A broad ruling in favor of the NFL could rewrite almost all of the sports antitrust law."





LEWIS SUSPENDED:

Rashard Lewis suspended without pay, losing \$1.6 MM of his \$18 MM 2009 10 contract. The All Star at 29 and a leading 3 point shooter in the NBA, tested positive for an over the counter supplement that contained DHEA, a steroids precursor, which elevated his testosterone level resulting in the Orlando Magic forward being suspended for the first 10 games of the 2009 10 regular season.

STALLWORTH SUSPENDED:

Donte Stallworth pled guilty to driving drunk. After pleading guilty to the charge of DUI manslaughter, after the car Stallworth was driving killed a pedestrian. Facing 15 years of prison time, Stallworth received a 30 day jail sentence and then reached a settlement with the family of the 59 year old construction worker killed by Stallworth from the driver's seat of his black 2005 Bentley. Stallworth also must serve 2 years of house arrest and spend 8 years on probation. After the sentence was handed down NFL Commission Roger Goodell suspended the Cleveland Browns receiver indefinitely!

CLARETT NOT RELEASED:

LEGAL POSTS

not seen in the Post Office

Maurice Clarett requested early release from prison. The DA in Columbus, Ohio opposes early release. Franklin County Prosecutor Ron O'Brien, says "Clarett hasn't proven that he should be released so early. Clarett, a former football star at Ohio State, is serving a minimum 3 1/2 years sentence for a holdup and subsequent highway chase. The police found loaded guns in the getaway car.

HALL NOT RELEASED:

Mel Hall former major league baseball player, got 45 years from a Tarrant County jury in Fort Worth, Texas. The 13 year career former ball player was accused of raping a 12 year old girl. The girl was a former player for Hall when he coached an elite girls basketball team. Hall was convicted on 3 counts of aggravated sexual assault and two counts of indecency with a child.

MAYO TOLD TO PAY:

O.J. Mayo, former basketball player at Southern Cal, is accused of ordering \$150,000 in jewelry and not paying for it. In the breach of contract lawsuit against the current Memphis Grizzlies player, the custom jewelry plaintiff complained that Mayo ordered jewelry such as diamond bangles, diamond rings, diamond necklaces, watches and bracelets. The lawsuit also alleges that Mayo induced the plaintiff to allow him to take the jewelry by promising to order more in future years, "as much as \$150,000 per year." Instead, Mayo never paid for the custom made jewelry he previously ordered, according to the lawsuit in Los Angeles Superior Court.

SAVE THE DATE

OCTOBER 1-2, 2009 The 19th Annual Entertainment Law Institute

Hyatt Regency Hotel • Austin, Texas

This is the same weekend as the AUSTIN CITY LIMITS MUSIC FESTIVAL which starts on Friday.

SUBMIT YOUR ARTICLES

The editors of the *Texas Entertainment and Sports Law Journal ("Journal")* are soliciting articles on a sports or entertainment law topic for publication in the TESLAW Journal.

All submitted articles will be considered for publication in the *Journal*. Although all submitted articles may not be published, we may choose to publish more than one article to fulfill our mission of providing current practical and scholarly literature to Texas lawyers practicing sports or entertainment law.

All articles should be submitted to the editor and conform to the following general guidelines. Articles submitted for publication in the Spring 2010 issue of the journal must be received no later than January 1, 2010.

<u>Length:</u> no more than twenty-five typewritten, double-spaced pages, including any endnotes. Space limitations usually prevent us from publishing articles longer in length.

<u>Endnotes:</u> must be concise, placed at the end of the article, and in Harvard "Blue Book" or Texas Law Review "Green Book" form.

<u>Form</u>: typewritten, double-spaced on $8\frac{1}{2}$ " x 11" paper and submitted in triplicate with a diskette indicating its format.

If you have any questions concerning the *Journal*, please email Sylvester R, Jaime, Editor, Texas Entertainment & Sports Law Journal, at srjaimelaw@ comcast.net.

NOTICE:

Art-friendly journal seeking budding artist to display artwork on cover! If you would like to see your (or your client's, mother's, spouse's, friend's, etc.) artwork on the cover of our journal, please submit a JPEG or EPS file (no less than 300 dpi) along with a PDF of the artwork to Sylvester Jaime at *srjaimelaw@comcast.net*.

Articles appearing in the Journal are selected for content and subject matter. Readers should assure themselves that the material contained in the articles is current and applicable to their needs. Neither the Section nor the Journal Staff warrant the material to be accurate or current. Readers should verify statements and information before relying on them. If you become aware of inaccuracies, new legislation, or changes in the law as used, please contact the Journal Editor. The material appearing in the Journal is not a substitute for competent independent legal advice. **RECENT CASES OF INTEREST** Prepared by the South Texas College of Law Students South Texas College of Sports Law & Entertainment Society

COWBOYS WIN TRADEMARK DISPUTE: "AMERICA'S TEAM" BELONGS TO THEM

The Dallas Cowboys recently won a trademark infringement case that protected its use of the phrase "America's Team" even though it had been previously trademarked by another entity, America's Team Properties (ATP). Dallas Cowboys Football Club and NFL Properties v. America's Team Properties, 616 F.Supp. 2d 622 (N.D. Tex. 2009).

The dispute arose after two parties claimed "trademark priority" over the phrase "America's Team". In 1979, when they released their 1978 season highlight film, the Cowboys began using the phrase "America's Team". For the next decade, the Cowboys repeatedly used the phrase in commerce while encouraging sponsors and other corporate entities to use the phrase when referring to the Cowboys. More specifically, from 1979 until 1990, the Cowboys sold merchandise that identified them as "America's Team" and used the term in publications, multimedia, and other team-oriented paraphernalia. In 1991, the team began selling "America's Team" t-shirts. Finally, in 1992, the Cowboys procured a Texas trademark for the phrase "America's Team."

In 1990, two years prior to the Cowboys registering of the trademark in Texas, a Minnesota corporation (ATP) registered the trademark "America's Team" and used the phrase on clothing. Then, in 1999, ATP ran an advertisement in the USA Today announcing an auction for the trademark, with a starting bid of \$500,000. At that time, ATP contacted and notified the Cowboys about the impending auction. In 2003, after nobody was willing to pay \$500,000 for the trademark, ATP again contacted the Cowboys and offered to sell them the trademark for \$400,000. After refusing to do so, the Cowboys brought suit against ATP and sought to enjoin them from using the phrase "America's Team". The Cowboys brought suit for trademark infringement and based the claim on state and common law trademark principles. The Cowboys claimed that they had "trademark priority" over the phrase "America's Team" even though ATP had federally registered the phrase in 1990 and the Cowboys had not registered it until 1992.

The district court first noted that the party who first registers a trademark has presumptive ownership of that trademark. In the current case, ATP had presumptive ownership of the "America's Team" trademark because they were the first party to register the phrase. Another party, however, can overcome this presumption of ownership by proving that they had previously used the trademark. According to the district court, "federal registration does not cut off the pre existing common law rights of others." In other words, the registration of a mark at the federal level does not wipe out the prior, nonregistered, common law rights of others. Thus, a party who uses an unregistered mark for a period of time can establish a common law trademark and have priority over another party who subsequently registers the mark. In the current case, the Cowboys had "trademark priority" because they had used the phrase "America's Team" for more than a decade prior to ATP's registration of the phrase. As a result, even though ATP had registered the trademark prior to the Cowboys' registration, the court enjoined ATP's use of "America's Team".

By: Bradley V. Rochman, South Texas College of Law

The New and Updated Section Website is at www.teslaw.org. Comments or suggestions may be submitted to Kenneth W. Prajak at ken@bannerot.com or your editor at srjaimelaw@comcast.net ...

OFFICE SPACE AVAILABLE IN AUSTIN 2106 E. M.L.K. Blvd.

ENTERTAINMENT ATTORNEY PREFERRED

CONTACT: MIKE TOLLESON

mike@miketolleson.com subject: Office Space

RECENT SPORTS AND ENTERTAINMENT LAW PUBLICATIONS

Compiled by Monica Ortale, Faculty Services & Reference Librarian The Fred Parks Law Library, South Texas College of Law

SPORTS LAW BIBLIOGRAPHY: AMATEUR SPORTS

Wes R. McCart. Note. *Title IX Crashes the Party in College Athletic Recruiting*. (Simpson v. Univ. of Colo., 372 F. Supp. 2d 1229, 2005.) 58 DEPAUL L. REV. 153 (2008).

Ann Scales. Student Gladiators And Sexual Assault: A New Analysis Of Liability For Injuries Inflicted By College Athletes, 15 MICH. J. GENDER & L. 205 (2009).

ANTITRUST

Justice Samuel A. Alito, Jr. *The Origin Of The Baseball Antitrust Exemption:* Federal Baseball Club Of Baltimore, Inc. v. National League Of Professional Baseball Clubs, 34 J. SUP. CT. HIST. 183 (2009).

Ethan Flatt. Note. *Solidifying The Defensive Line: The NFL Network's Current Position Under Antitrust Law And How It Can Be Improved*, 11 VAND. J. ENT. & TECH. L. 637 (2009).

ARBITRATION

Abbas Ravjani. The Court Of Arbitration For Sport: A Subtle Form Of International Delegation, 2 J. INT'L MEDIA & ENT. L. 241 (2009).

CONSTITUTIONAL LAW

Paul T. Derousselle II. Comment. *Personal Foul! How The Supreme Court Allowed The Tennessee Secondary School Athletic Association To Violate Brentwood Academy s First Amendment Rights*, 36 S.U. L. REV. 173 (2008).

Christopher J. Kaufman. Comment. Unsportsmanlike Conduct: 15-Yard Penalty And Loss Of Free Speech In Public University Sports Stadiums, 57 U. KAN. L. REV. 1235 (2009).

CONTRACTS

Sara Young. Note. Does A Coach Owe Players A Fiduciary Duty? Examining The Relationship Between Coach And Team, 35 J.C.& U.L. 475 (2009).

DISCRIMINATION

Christopher Bidlack. Comment. The Prohibition Of Prosthetic Limbs In American Sports: The Issues And The Role Of The Americans With Disabilities Act, 19 MARQ. SPORTS L. REV. 613 (2009).

Storms, Jeffrey S. El Montículo (AThe Mound@): The Disparate Treatment Of Latin American Baseball Players In Major League Baseball, 11 U. ST. THOMAS J.L. & PUB. POL'Y. 81 (2009).

GENDER

Erin Denatale. Comment. FORE: *The Problem With The LPGA's Proposed Language* Policy, 30 WHITTIER L. REV. 623 (2009).

INTELLECTUAL PROPERTY

Daniel Bilsky. From Parts Unknown: WWEv. Jim Hellwig In The Ultimate Battle For Character Copyright, 19 MARQ. SPORTS L. REV. 419 (2009).

INTERNATIONAL

Anita L. DeFrantz. *Which Rules?: International Sport and Doping in the* 21st Century, 31 HOUS. J. INT'L L. 1 (2008).

LABOR

Matthew Epps. Comment. *Full Court Press: How Collective Bargaining Weakened the NBA's Competitive Edge in a Globalized Sport*, 16 VILL. SPORTS & ENT. L.J. 343 (2009).

MEDICAL

Andrew D. Hohenstein, Comment. *Team Physicians: Adhering To The Hippocratic Oath Or Just Plain Hypocrites*? 19 MARQ. SPORTS L. REV. 579 (2009).

MISCELLANEOUS

Erick S. Lee. Play Ball!: Substituting Current Federal Non-Regulation Of Fantasy Sports Leagues With Limited Supervision Of Hyper-Competitive Leagues, 29 LOY. L.A. ENT. L. REV. 53 (2008-2009).

PROFESSIONAL SPORTS

Marc Edelman & Brian Doyle. Antitrust And AFree Movement@ Risks Of Expanding U.S. Professional Sports Leagues Into Europe, 29 NW. J. INT'L L. & BUS. 403 (2009).

Holli N. Heiles. Comment. Baseball's AGrowth@ Problem: Can Congress Require Major League Baseball To Test Its Athletes For Human Growth Hormone? A Proposal, 62 ARK. L. REV. 315 (2009).

Sarah R. Heisler. Note. *Steroid Regulation In Professional Sports: Sarbanes-Oxley As A Guide*, 27 CARDOZO ARTS & ENT. L.J. 199 (2009).

Rick J. Lopez. Comment. Signing Bonus Skimming And A Premature Call For A Global Draft In Major League Baseball, 41 ARIZ. ST. L.J. 349 (2009).

PROPERTY

Steinkohl, Bryan. Comment. Now Taking The Field, The State Government: Landmark Status Of Baseball Stadiums As Regulatory Takings, 29 LOY. L.A. ENT. L. REV. 233 (2009).

SYMPOSIA

Symposium: Doping In Sports: Legal And Ethical Issues, 19 MARQ. SPORTS L. REV. 1 (2008).

TAX

Michael Halper. 2008 Note Of The Year. *The Tax Ramifications Of Catching Home Run Baseballs*. 59 CASE W. RES. L. REV. 191 (2008).

Robert Holo, & Jonathan Talansky. *Taxing The Business Of Sports*, 9 FLA. TAX REV. 161 (2008).

TORTS

Evan Steele Fensterstock. Comment. Shim v. Ahn *Applies The Primary Assumption Of Risk Doctrine To Injuries Sustained By Golfers In The Same Group: Negligence Goes Unpunished*, 43 NEW ENG. L. REV. 87 (2008).

Brian P. Harlan. Note. *The California Supreme Court Should Take A Mulligan: How The Court Shanked By Applying The Primary Assumption Of Risk Doctrine To Golf.* (Shin v. Ahn, 165 P.3d 581, 2007.) 29 LOY. L.A. ENT. L. REV. 91 (2008-2009).

ENTERTAINMENT LAW BIBLIOGRAPHY: ART

Denise M. Alter. Corporate Art Collecting And Fiduciary Duties To Shareholders: Legal Duties And Best Practices For Directors And Officers, 2009 COLUM. BUS. L. REV. 1.

Continued from page 22

Hilary Jay. Note. *Picture Imperfect: the Rights of Art Consignor-Collectors When Their Art Dealer Files for Bankruptcy*, 58 DUKE L.J. 1859-1893 (2009).

Anna Belle Wilder Norton. Comment. Site-Specific Art Gets A Bum Wrap: Illustrating The Limitations Of The Visual Artists Rights Act Of 1990 Through A Study Of Christo And Jeanne-Claude's Unique Art., 39 CUMB. L. REV. 749 (2008-2009).

Maritza . Schaeffer. Note. Contemporary Issues In The Visual Art Realm: How Useful Are Creative Commons Licenses? 17 J.L. & POL'Y 359 (2008).

COPYRIGHT

Michael Katz. Recycling Copyright: Survival & Growth In The Remix Age, 13 INTELL. PROP. L. BULL. 21 (2008).

Dotan Oliar & Christopher Sprigman. There's No Free Laugh (Anymore): The Emergence Of Intellectual Property Norms And The Transformation Of Stand-Up Comedy, 94 VA. L. REV. 1787 (2008).

Robert Platt. Comment. The Family Movie Act And Its Effect On U.S. Copyright Law, 40 U. TOL. L. REV. 487 (2009).

Meg Reid. Casenote. *Harry Potter and the Copyright Act: Have The Courts Finally Waved A Magic Wand By Defining Fair Use For Secondary Authors?* (Warner Bros., Entm't, Inc. v. RDR Books, 575 F. Supp. 2d 513, 2008.) 16 VILL. SPORTS & ENT. L.J. 291 (2009).

Maureen Steimer. Casenote. *Restoring the Balance: Bringing Back Consumer Rights* In UMG Recordings v. Augusto *by Reaffirming the First Sale Doctrine in Copyright Law.* (UMG Recordings v. Augusto, 558 F. Supp. 2d 1055, 2008.) 16 VILL. SPORTS & ENT. L.J. 313 (2009).

CRIMINAL

Derek R. Kelly. Note. Illegal Tender: Antiquities Protection And U.S. Import Restrictions On Cypriot Coinage, 34 BROOK. J. INT'L L. 491 (2009).

M.J. Williams. Note. Framing Art Vandalism: A Proposal To Address Violence Against Art, 74 BROOK. L. REV. 581 (2009).

FAIR USE

Christopher Reid. Note. *Fair Game: The Application of the Fair Use Doctrine to Machinima.* 19 FORDHAM INTELL. PROP. MEDIA & ENT. L.J. 831 (2009).

INTERNATIONAL

Edythe E. Huang. Note. Looting In Iraq, Five Years Later: An Evaluation Of The International Protection, Recovery, And Repatriation Of Looted Cultural Artifacts, 10 RUTGERS RACE & L. REV. 183 (2008).

Alexander Mackintosh Ritchie. Student Article. Victorious Youth In Peril: Analyzing Arguments Used In Cultural Property Disputes To Resolve The Case Of The Getty Bronze, 9 PEPP. DISP. RESOL. L.J. 325 (2009).

Katherine D. Vitale. Note. *The War On Antiquities: United States Law* And Foreign Cultural Property, 84 NOTRE DAME L. REV. 1835 (2009).

Yong Wan. Legal Protection Of Performers' Rights In The Chinese Copyright Law, 56 J. COPYRIGHT SOC'Y U.S.A. 669 (2009).

MISCELLANEOUS

Edwin M. McPherson, *Did Marathon And Preston Kill The Talent Agencies Act?* 38 SW. L. REV. 443 (2009).

Paul Riley. Note. *Litigating Second Life Land Disputes: A Consumer Protection Approach*, 19 FORDHAM INTELL. PROP. MEDIA & ENT. L.J. 877 (2009).

Alisa B. Steinberg. Note. For Sale--One Level 5 Barbarian for 94,800 Won: The International Effects of Virtual Property and the Legality of Its Ownership, 37 GA. J. INT'L & COMP. L. 381 (2009).

MUSIC

Robert Brauneis. *Copyright And The World's Most Popular Song*, 56 J. COPYRIGHT SOC'Y U.S.A. 335 (2009).

Keith Harris. Note. For Promotional Use Only: Is The Resale Of A Promotional CD Protected By The First Sale Doctrine? 30 CARDOZO L. REV. 1745 (2009).

Sara Karubian. Note. 3601 Deals: An Industry Reaction to the Devaluation of Recorded Music. 18 S. CAL. INTERDISC. L.J. 395 (2009).

Mark F. Schultz. *Live Performance, Copyright, and the Future of the Music Business*, 43 U. RICH. L. REV. 685 (2009).

PROFESSIONAL RESPONSIBILITY

Jennifer Anglim. Kreder, The Holocaust, Museum Ethics And Legalism, 18 S. CAL. REV. L. & SOC. JUST. 1 (2008).

SYMPOSIA

Symposium. Media, Justice, And The Law, 61 STAN. L. REV. 1333-1533 (2009).

Symposium. Publicity Rights In Bytes: Contemporary Issues In Entertainment And Sports Law, 11 CHAPMAN L. REV. 423 (2008).

Symposium. James Bessen And Michael J. Meurer's *Patent Failure: How Judges, Bureaucrats, And Lawyers Put Innovators At Risk.* 16 J. INTELL. PROP. L. 197 (2009).

Symposium. *The House That Tax Payers Built: Stadiums, Speech and Public Funding,* 16 VILL. SPORTS & ENT. L.J. 255 (2009).

TAX

Steven Chung. Note. Real *Taxation Of Virtual Commerce*, 28 VA. TAX REV. 733 (2009).

Theodore P. Seto. *When is a Game Only a Game?: the Taxation of Virtual Worlds*, 77 U. CIN. L. REV. 1027 (2009).

Stephen Taylor. Comment. "Are You Not Entertained? Is This Not Why You Are Here?" U.S. Taxation of Foreign Athletes and Entertainers, 16 VILL. SPORTS & ENT. L.J. 375 (2009).

TELEVISION

Adam P. Greenberg. Note. *Reality's Kids: Are Children Who Participate On Reality Television Shows Covered Under The Fair Labor Standards Act?* 82 S. CAL. L. REV. 595 (2009).

Elena Marty-Nelson. *Securities Laws In Soap Operas And Telenovelas:* Are All My Children Engaged In Securities Fraud? 18 S. CAL. INTERDISC. L.J. 329 (2009).

THEATRE

Margit Livingston. Inspiration Or Imitation: Copyright Protection For Stage Directions, 50 B.C. L. REV. 427 (2009).

TORTS

Timothy Dylan Reeves. Note. *Tort Liability For Manufacturers Of Violent Video Games: A Situational Discussion Of The Causation Calamity*, 60 ALA. L. REV. 519 (2009).

TRADEMARK

Melissa Ung. Comment. Trademark Law and the Repercussions of Virtual Property (IRL). 17 COMMLAW CONSPECTUS 679 (2009).

State Bar of Texas P. O. Box 12487 Austin, TX 78711-2487

NON PROFIT ORGANIZATION U.S. POSTAGE **PAID** PERMIT NO. 1804 AUSTIN, TEXAS

Join the Entertainment and Sports Law Section

Formed in 1989, the Entertainment and Sports Law Section has over 500 members. The Section is directed at lawyers who devote a portion of their practice to entertainment and/or sports law and seeks to educate its members on recent developments in entertainment and sports law. Membership in the Section is also available to non-lawyers who have an interest in these areas.

The Entertainment & Sports Law Journal, published twice a year by the Section, contains articles and information of professional and academic interest relating to entertainment, sports, intellectual property, art, and related areas. The Section also cosponsors seminars of interest to its members, such as this annual Entertainment Law Institute.



Membership in the Section is from June 1 to May 31. To join, go to TexasBar.com and login to MyBarPage. There you will see "My Sections" and a chance to click on "Purchase Sections." Check "Entertainment & Sports Law" and pay the \$30 annual fee with a credit card. Or complete the information below and forward it with a check in the amount of \$30.00 (made payable to ENTERTAINMENT & SPORTS LAW SECTION) to Michele Schweitzer, P.O. Box 12487, Capitol Station, Austin, TX 78711.

NAME____

ADDRESS

BAR CARD NO.